% This TeX file was generated by /usr/local/bin/texpsinclude
% from the original TeX file generic.tex
% and the PostScript files bump1.eps homfig2.eps homfig2.fig.
% TeX writes out the included PostScript to files.
%
% Here is part of generic.tex:

\documentstyle[12pt,amssymb]{amsart}

% Here are the
% TeX macros for dumping included Postscript to files.
% Adapted from Knuth's \answer macro in the TeXbook.
% Jamie Stephens, jamies@math.utexas.edu, 28 Nov 94

\def\endofps{EndOfTheIncludedPostscriptMagicCookie}
\chardef\other=12
\newwrite\psdumphandle 
\outer\def\psdump#1{\par\medbreak
  \immediate\openout\psdumphandle=#1
  \copytoblankline}
\def\copytoblankline{\begingroup\setupcopy\copypsline}
\def\setupcopy{\def\do##1{\catcode`##1=\other}\dospecials
  \catcode`\\=\other \obeylines}
{\obeylines \gdef\copypsline#1
  {\def\next{#1}%
  \ifx\next\endofps\let\next=\endgroup %
  \else\immediate\write\psdumphandle{\next} \let\next=\copypsline\fi\next}}
\outer\def\closepsdump{
  \immediate\closeout\psdumphandle}

% Here is the PostScript for bump1.eps:
\message{Writing file bump1.eps}
\psdump{bump1.eps}%!PS-Adobe-2.0 EPSF-2.0
%%Title: homfig
%%Creator: fig2dev Version 3.1 Patchlevel 1
%%CreationDate: Wed Jan 28 14:26:57 1998
%%For: paternag@sol2 (PATERNAIN GABRIEL PEDRO ,05/03/1998)
%%Orientation: Landscape
%%BoundingBox: 0 0 378 407
%%Pages: 0
%%BeginSetup
%%IncludeFeature: *PageSize Letter
%%EndSetup
%%EndComments
/$F2psDict 200 dict def
$F2psDict begin
$F2psDict /mtrx matrix put
/col-1 {} def
/col0 {0.000 0.000 0.000 srgb} bind def
/col1 {0.000 0.000 1.000 srgb} bind def
/col2 {0.000 1.000 0.000 srgb} bind def
/col3 {0.000 1.000 1.000 srgb} bind def
/col4 {1.000 0.000 0.000 srgb} bind def
/col5 {1.000 0.000 1.000 srgb} bind def
/col6 {1.000 1.000 0.000 srgb} bind def
/col7 {1.000 1.000 1.000 srgb} bind def
/col8 {0.000 0.000 0.560 srgb} bind def
/col9 {0.000 0.000 0.690 srgb} bind def
/col10 {0.000 0.000 0.820 srgb} bind def
/col11 {0.530 0.810 1.000 srgb} bind def
/col12 {0.000 0.560 0.000 srgb} bind def
/col13 {0.000 0.690 0.000 srgb} bind def
/col14 {0.000 0.820 0.000 srgb} bind def
/col15 {0.000 0.560 0.560 srgb} bind def
/col16 {0.000 0.690 0.690 srgb} bind def
/col17 {0.000 0.820 0.820 srgb} bind def
/col18 {0.560 0.000 0.000 srgb} bind def
/col19 {0.690 0.000 0.000 srgb} bind def
/col20 {0.820 0.000 0.000 srgb} bind def
/col21 {0.560 0.000 0.560 srgb} bind def
/col22 {0.690 0.000 0.690 srgb} bind def
/col23 {0.820 0.000 0.820 srgb} bind def
/col24 {0.500 0.190 0.000 srgb} bind def
/col25 {0.630 0.250 0.000 srgb} bind def
/col26 {0.750 0.380 0.000 srgb} bind def
/col27 {1.000 0.500 0.500 srgb} bind def
/col28 {1.000 0.630 0.630 srgb} bind def
/col29 {1.000 0.750 0.750 srgb} bind def
/col30 {1.000 0.880 0.880 srgb} bind def
/col31 {1.000 0.840 0.000 srgb} bind def

end
save
-29.0 -18.0 translate
 90 rotate
1 -1 scale

/clp {closepath} bind def
/ef {eofill} bind def
/gr {grestore} bind def
/gs {gsave} bind def
/l {lineto} bind def
/m {moveto} bind def
/n {newpath} bind def
/s {stroke} bind def
/slc {setlinecap} bind def
/slj {setlinejoin} bind def
/slw {setlinewidth} bind def
/srgb {setrgbcolor} bind def
/rot {rotate} bind def
/sc {scale} bind def
/tr {translate} bind def
/tnt {dup dup currentrgbcolor
  4 -2 roll dup 1 exch sub 3 -1 roll mul add
  4 -2 roll dup 1 exch sub 3 -1 roll mul add
  4 -2 roll dup 1 exch sub 3 -1 roll mul add srgb}
  bind def
/shd {dup dup currentrgbcolor 4 -2 roll mul 4 -2 roll mul
  4 -2 roll mul srgb} bind def
 /DrawEllipse {
/endangle exch def
/startangle exch def
/yrad exch def
/xrad exch def
/y exch def
/x exch def
/savematrix mtrx currentmatrix def
x y tr xrad yrad sc 0 0 1 startangle endangle arc
closepath
savematrix setmatrix
} def

 /DrawSplineSection {
/y3 exch def
/x3 exch def
/y2 exch def
/x2 exch def
/y1 exch def
/x1 exch def
/xa x1 x2 x1 sub 0.666667 mul add def
/ya y1 y2 y1 sub 0.666667 mul add def
/xb x3 x2 x3 sub 0.666667 mul add def
/yb y3 y2 y3 sub 0.666667 mul add def
x1 y1 lineto
xa ya xb yb x3 y3 curveto
} def

/$F2psBegin {$F2psDict begin /$F2psEnteredState save def} def
/$F2psEnd {$F2psEnteredState restore end} def
%%EndProlog

$F2psBegin
10 setmiterlimit
 0.06000 0.06000 sc
15.000 slw
% Ellipse
n 3450 1350 420 420 0 360 DrawEllipse gs col0 s gr

% Ellipse
n 3450 3150 450 450 0 360 DrawEllipse gs col0 s gr

% Ellipse
n 3450 5925 450 450 0 360 DrawEllipse gs col0 s gr

30.000 slw
% Polyline
n 5475 6750 m 5475 5025 l  1800 5025 l  1800 6750 l  clp  gs col0 s gr
7.500 slw
% Polyline
n 1350 2100 m 2700 1500 l  2700 1500 l  gs col0 s gr
% Polyline
n 1350 2100 m 2550 3000 l  gs col0 s gr
% Polyline
n 6900 3225 m 6900 5175 l  gs col0 s gr
n 6937.50 5040.00 m 6900.00 5175.00 l 6862.50 5040.00 l gs col0 s gr
% Polyline
n 1500 6000 m 2775 6000 l  gs col0 s gr
30.000 slw
% Polyline
n 1725 4050 m 5400 4050 l  5400 525 l  1725 525 l  clp  gs col-1 s gr
15.000 slw
% Open spline
n 3375.00 2625.00 m 3262.50 2662.50 l
3262.50 2662.50 3150.00 2700.00 3037.50 2850.00 DrawSplineSection
3037.50 2850.00 2925.00 3000.00 2925.00 3112.50 DrawSplineSection
2925.00 3112.50 2925.00 3225.00 3000.00 3412.50 DrawSplineSection
3000.00 3412.50 3075.00 3600.00 3225.00 3637.50 DrawSplineSection
3225.00 3637.50 3375.00 3675.00 3562.50 3675.00 DrawSplineSection
3562.50 3675.00 3750.00 3675.00 3900.00 3562.50 DrawSplineSection
3900.00 3562.50 4050.00 3450.00 4162.50 3262.50 DrawSplineSection
4162.50 3262.50 4275.00 3075.00 4387.50 2812.50 DrawSplineSection
4387.50 2812.50 4500.00 2550.00 4575.00 2287.50 DrawSplineSection
4575.00 2287.50 4650.00 2025.00 4612.50 1800.00 DrawSplineSection
4612.50 1800.00 4575.00 1575.00 4462.50 1350.00 DrawSplineSection
4462.50 1350.00 4350.00 1125.00 4125.00 975.00 DrawSplineSection
4125.00 975.00 3900.00 825.00 3712.50 787.50 DrawSplineSection
3712.50 787.50 3525.00 750.00 3375.00 787.50 DrawSplineSection
3375.00 787.50 3225.00 825.00 3112.50 937.50 DrawSplineSection
3112.50 937.50 3000.00 1050.00 2962.50 1237.50 DrawSplineSection
2962.50 1237.50 2925.00 1425.00 3037.50 1612.50 DrawSplineSection
3037.50 1612.50 3150.00 1800.00 3337.50 1837.50 DrawSplineSection
3337.50 1837.50 3525.00 1875.00 3675.00 1762.50 DrawSplineSection
3675.00 1762.50 3825.00 1650.00 3825.00 1650.00 DrawSplineSection
3825.00 1650.00 l  gs col-1 s gr

[100.0] 0 setdash
% Open spline
n 3225.00 2925.00 m 3225.00 2925.00 l
3225.00 2925.00 3225.00 2925.00 3225.00 2925.00 DrawSplineSection
3225.00 2925.00 l  gs col-1 s gr
[] 0 setdash
[100.0] 0 setdash
% Open spline
n 3075.00 3000.00 m 3075.00 3000.00 l
3075.00 3000.00 l  gs col-1 s gr
[] 0 setdash
[100.0] 0 setdash
% Open spline
n 3150.00 2925.00 m 3262.50 2850.00 l
3262.50 2850.00 3375.00 2775.00 3450.00 2775.00 DrawSplineSection
3450.00 2775.00 3525.00 2775.00 3637.50 2850.00 DrawSplineSection
3637.50 2850.00 3750.00 2925.00 3787.50 3000.00 DrawSplineSection
3787.50 3000.00 3825.00 3075.00 3825.00 3187.50 DrawSplineSection
3825.00 3187.50 3825.00 3300.00 3750.00 3375.00 DrawSplineSection
3750.00 3375.00 3675.00 3450.00 3562.50 3487.50 DrawSplineSection
3562.50 3487.50 3450.00 3525.00 3337.50 3450.00 DrawSplineSection
3337.50 3450.00 3225.00 3375.00 3187.50 3337.50 DrawSplineSection
3187.50 3337.50 3150.00 3300.00 3000.00 3187.50 DrawSplineSection
3000.00 3187.50 2850.00 3075.00 2775.00 2925.00 DrawSplineSection
2775.00 2925.00 2700.00 2775.00 2625.00 2625.00 DrawSplineSection
2625.00 2625.00 2550.00 2475.00 2512.50 2212.50 DrawSplineSection
2512.50 2212.50 2475.00 1950.00 2475.00 1762.50 DrawSplineSection
2475.00 1762.50 2475.00 1575.00 2512.50 1387.50 DrawSplineSection
2512.50 1387.50 2550.00 1200.00 2625.00 1012.50 DrawSplineSection
2625.00 1012.50 2700.00 825.00 2812.50 787.50 DrawSplineSection
2812.50 787.50 2925.00 750.00 3037.50 787.50 DrawSplineSection
3037.50 787.50 3150.00 825.00 3225.00 862.50 DrawSplineSection
3225.00 862.50 3300.00 900.00 3450.00 975.00 DrawSplineSection
3450.00 975.00 3600.00 1050.00 3675.00 1125.00 DrawSplineSection
3675.00 1125.00 3750.00 1200.00 3787.50 1312.50 DrawSplineSection
3787.50 1312.50 3825.00 1425.00 3787.50 1500.00 DrawSplineSection
3787.50 1500.00 3750.00 1575.00 3675.00 1650.00 DrawSplineSection
3675.00 1650.00 3600.00 1725.00 3450.00 1725.00 DrawSplineSection
3300.00 1725.00 l  gs col-1 s gr
[] 0 setdash
/Helvetica-Bold findfont 300.00 scalefont setfont
6750 2775 m
gs 1 -1 sc (Mo) col0 show gr
/Helvetica-Bold findfont 300.00 scalefont setfont
6900 5850 m
gs 1 -1 sc (M) col0 show gr
/Times-Italic findfont 300.00 scalefont setfont
6600 4050 m
gs 1 -1 sc (p) col0 show gr
/Times-Italic findfont 300.00 scalefont setfont
-450 5700 m
gs 1 -1 sc (Projection of the) col0 show gr
/Times-Italic findfont 300.00 scalefont setfont
-450 5970 m
gs 1 -1 sc (static class of L) col0 show gr
/Times-Italic findfont 300.00 scalefont setfont
-450 1875 m
gs 1 -1 sc (Projection of the) col0 show gr
/Times-Italic findfont 300.00 scalefont setfont
-450 2415 m
gs 1 -1 sc (of Lo) col0 show gr
/Times-Italic findfont 300.00 scalefont setfont
-450 2145 m
gs 1 -1 sc (static classes) col0 show gr
$F2psEnd
restore


EndOfTheIncludedPostscriptMagicCookie

\closepsdump

% Here is the PostScript for homfig2.eps:
\message{Writing file homfig2.eps}
\psdump{homfig2.eps}%!PS-Adobe-2.0 EPSF-2.0
%%Title: homfig2.eps
%%Creator: fig2dev Version 3.2 Patchlevel 0-beta3
%%CreationDate: Wed Nov 18 20:09:06 1998
%%For: gonzalo@pibil.cimat.mx (RHS Linux User)
%%Orientation: Portrait
%%BoundingBox: 0 0 413 317
%%Pages: 0
%%BeginSetup
%%EndSetup
%%Magnification: 1.0000
%%EndComments
/$F2psDict 200 dict def
$F2psDict begin
$F2psDict /mtrx matrix put
/col-1 {0 setgray} bind def
/col0 {0.000 0.000 0.000 srgb} bind def
/col1 {0.000 0.000 1.000 srgb} bind def
/col2 {0.000 1.000 0.000 srgb} bind def
/col3 {0.000 1.000 1.000 srgb} bind def
/col4 {1.000 0.000 0.000 srgb} bind def
/col5 {1.000 0.000 1.000 srgb} bind def
/col6 {1.000 1.000 0.000 srgb} bind def
/col7 {1.000 1.000 1.000 srgb} bind def
/col8 {0.000 0.000 0.560 srgb} bind def
/col9 {0.000 0.000 0.690 srgb} bind def
/col10 {0.000 0.000 0.820 srgb} bind def
/col11 {0.530 0.810 1.000 srgb} bind def
/col12 {0.000 0.560 0.000 srgb} bind def
/col13 {0.000 0.690 0.000 srgb} bind def
/col14 {0.000 0.820 0.000 srgb} bind def
/col15 {0.000 0.560 0.560 srgb} bind def
/col16 {0.000 0.690 0.690 srgb} bind def
/col17 {0.000 0.820 0.820 srgb} bind def
/col18 {0.560 0.000 0.000 srgb} bind def
/col19 {0.690 0.000 0.000 srgb} bind def
/col20 {0.820 0.000 0.000 srgb} bind def
/col21 {0.560 0.000 0.560 srgb} bind def
/col22 {0.690 0.000 0.690 srgb} bind def
/col23 {0.820 0.000 0.820 srgb} bind def
/col24 {0.500 0.190 0.000 srgb} bind def
/col25 {0.630 0.250 0.000 srgb} bind def
/col26 {0.750 0.380 0.000 srgb} bind def
/col27 {1.000 0.500 0.500 srgb} bind def
/col28 {1.000 0.630 0.630 srgb} bind def
/col29 {1.000 0.750 0.750 srgb} bind def
/col30 {1.000 0.880 0.880 srgb} bind def
/col31 {1.000 0.840 0.000 srgb} bind def

end
save
-61.0 335.0 translate
1 -1 scale

/cp {closepath} bind def
/ef {eofill} bind def
/gr {grestore} bind def
/gs {gsave} bind def
/sa {save} bind def
/rs {restore} bind def
/l {lineto} bind def
/m {moveto} bind def
/rm {rmoveto} bind def
/n {newpath} bind def
/s {stroke} bind def
/sh {show} bind def
/slc {setlinecap} bind def
/slj {setlinejoin} bind def
/slw {setlinewidth} bind def
/srgb {setrgbcolor} bind def
/rot {rotate} bind def
/sc {scale} bind def
/sd {setdash} bind def
/ff {findfont} bind def
/sf {setfont} bind def
/scf {scalefont} bind def
/sw {stringwidth} bind def
/tr {translate} bind def
/tnt {dup dup currentrgbcolor
  4 -2 roll dup 1 exch sub 3 -1 roll mul add
  4 -2 roll dup 1 exch sub 3 -1 roll mul add
  4 -2 roll dup 1 exch sub 3 -1 roll mul add srgb}
  bind def
/shd {dup dup currentrgbcolor 4 -2 roll mul 4 -2 roll mul
  4 -2 roll mul srgb} bind def
 /DrawEllipse {
/endangle exch def
/startangle exch def
/yrad exch def
/xrad exch def
/y exch def
/x exch def
/savematrix mtrx currentmatrix def
x y tr xrad yrad sc 0 0 1 startangle endangle arc
closepath
savematrix setmatrix
} def

/$F2psBegin {$F2psDict begin /$F2psEnteredState save def} def
/$F2psEnd {$F2psEnteredState restore end} def
%%EndProlog

$F2psBegin
10 setmiterlimit
n -1000 6573 m -1000 -1000 l 8888 -1000 l 8888 6573 l cp clip
 0.06000 0.06000 sc
45.000 slw
% Ellipse
n 2100 1800 900 900 0 360 DrawEllipse gs col0 s gr

% Ellipse
n 6900 1800 849 849 0 360 DrawEllipse gs col0 s gr

% Ellipse
n 4500 4500 900 900 0 360 DrawEllipse gs col0 s gr

% Polyline
7.500 slw
gs  clippath
4422 237 m 4854 327 l 4422 417 l 4896 417 l 4896 237 l cp
clip
n 4830 327 m 4881 327 l gs col0 s gr gr

% arrowhead
n 4422 237 m 4854 327 l 4422 417 l 4494 327 l 4422 237 l  cp gs 0.00 setgray ef gr  col0 s
% Polyline
gs  clippath
2318 4273 m 2094 3892 l 2451 4152 l 2133 3801 l 1999 3922 l cp
clip
n 2106 3906 m 2076 3873 l gs col0 s gr gr

% arrowhead
n 2318 4273 m 2094 3892 l 2451 4152 l 2336 4159 l 2318 4273 l  cp gs 0.00 setgray ef gr  col0 s
% Polyline
gs  clippath
7022 3848 m 6674 4119 l 6885 3731 l 6579 4093 l 6716 4210 l cp
clip
n 6690 4101 m 6657 4140 l gs col0 s gr gr

% arrowhead
n 7022 3848 m 6674 4119 l 6885 3731 l 6907 3845 l 7022 3848 l  cp gs 0.00 setgray ef gr  col0 s
% Polyline
gs  clippath
6094 1159 m 6327 980 l 6185 1238 l 6400 988 l 6309 909 l cp
clip
n 2794 1510 m 2794 1514 l 2795 1522 l 2796 1536 l 2797 1556 l 2798 1583 l
 2800 1614 l 2801 1648 l 2802 1685 l 2803 1721 l 2804 1755 l
 2804 1788 l 2804 1819 l 2803 1848 l 2802 1874 l 2800 1899 l
 2797 1922 l 2794 1944 l 2790 1965 l 2785 1986 l 2779 2006 l
 2773 2027 l 2766 2048 l 2758 2069 l 2749 2090 l 2740 2111 l
 2729 2133 l 2718 2155 l 2705 2176 l 2692 2198 l 2679 2219 l
 2665 2239 l 2650 2259 l 2635 2279 l 2620 2297 l 2605 2314 l
 2590 2331 l 2574 2346 l 2559 2361 l 2543 2376 l 2526 2390 l
 2510 2403 l 2492 2416 l 2474 2429 l 2455 2442 l 2435 2454 l
 2414 2466 l 2393 2478 l 2371 2489 l 2349 2499 l 2326 2509 l
 2304 2517 l 2281 2525 l 2259 2532 l 2236 2538 l 2215 2543 l
 2193 2547 l 2172 2550 l 2151 2553 l 2131 2555 l 2110 2555 l
 2089 2555 l 2067 2555 l 2046 2553 l 2023 2551 l 2000 2547 l
 1977 2543 l 1953 2538 l 1929 2533 l 1906 2526 l 1882 2519 l
 1859 2511 l 1836 2502 l 1814 2493 l 1793 2484 l 1772 2474 l
 1752 2464 l 1733 2454 l 1715 2443 l 1697 2432 l 1680 2421 l
 1663 2409 l 1646 2396 l 1629 2383 l 1612 2369 l 1595 2353 l
 1579 2337 l 1562 2321 l 1546 2303 l 1530 2285 l 1515 2266 l
 1501 2247 l 1487 2227 l 1474 2207 l 1462 2187 l 1451 2167 l
 1441 2147 l 1431 2127 l 1423 2107 l 1415 2086 l 1408 2067 l
 1401 2047 l 1395 2026 l 1390 2005 l 1385 1983 l 1380 1960 l
 1376 1936 l 1373 1912 l 1370 1887 l 1368 1862 l 1367 1836 l
 1366 1810 l 1366 1785 l 1367 1759 l 1369 1734 l 1371 1709 l
 1374 1685 l 1378 1662 l 1383 1639 l 1388 1616 l 1394 1595 l
 1401 1573 l 1407 1555 l 1414 1538 l 1422 1520 l 1430 1502 l
 1440 1483 l 1450 1464 l 1462 1445 l 1475 1426 l 1489 1406 l
 1504 1386 l 1521 1366 l 1539 1345 l 1558 1324 l 1578 1303 l
 1600 1282 l 1623 1260 l 1647 1239 l 1672 1218 l 1698 1196 l
 1725 1175 l 1753 1154 l 1782 1133 l 1811 1111 l 1842 1090 l
 1875 1069 l 1908 1047 l 1931 1032 l 1954 1018 l 1978 1003 l
 2003 987 l 2029 972 l 2056 956 l 2083 940 l 2112 924 l
 2142 907 l 2173 890 l 2205 873 l 2238 856 l 2273 838 l
 2308 821 l 2344 803 l 2382 785 l 2420 767 l 2459 749 l
 2499 732 l 2539 714 l 2581 696 l 2623 679 l 2665 662 l
 2708 645 l 2752 628 l 2795 612 l 2839 596 l 2883 581 l
 2928 566 l 2972 551 l 3017 538 l 3062 524 l 3106 511 l
 3151 499 l 3196 487 l 3241 475 l 3286 464 l 3331 454 l
 3377 444 l 3423 435 l 3463 427 l 3504 419 l 3545 412 l
 3587 405 l 3630 398 l 3673 392 l 3717 386 l 3762 380 l
 3808 375 l 3854 369 l 3901 364 l 3948 360 l 3997 355 l
 4046 351 l 4095 348 l 4146 344 l 4196 341 l 4248 338 l
 4299 336 l 4351 333 l 4403 332 l 4455 330 l 4508 329 l
 4560 328 l 4613 327 l 4665 327 l 4717 327 l 4768 327 l
 4820 327 l 4870 328 l 4921 329 l 4970 331 l 5020 332 l
 5068 334 l 5116 336 l 5163 338 l 5209 341 l 5255 344 l
 5299 347 l 5344 350 l 5387 353 l 5430 357 l 5472 360 l
 5513 364 l 5554 368 l 5595 373 l 5644 378 l 5692 384 l
 5740 390 l 5787 396 l 5835 402 l 5882 409 l 5930 417 l
 5977 424 l 6025 432 l 6072 441 l 6119 450 l 6166 459 l
 6213 468 l 6260 478 l 6306 488 l 6352 498 l 6398 509 l
 6443 520 l 6487 531 l 6531 543 l 6574 554 l 6616 566 l
 6657 578 l 6697 590 l 6736 602 l 6774 615 l 6811 627 l
 6847 639 l 6881 651 l 6915 664 l 6947 676 l 6978 688 l
 7008 701 l 7038 713 l 7066 725 l 7093 738 l 7119 750 l
 7145 763 l 7179 780 l 7212 797 l 7244 815 l 7275 833 l
 7305 851 l 7335 870 l 7364 890 l 7393 910 l 7421 930 l
 7448 951 l 7474 972 l 7499 993 l 7524 1015 l 7548 1037 l
 7570 1059 l 7592 1080 l 7612 1102 l 7631 1123 l 7649 1145 l
 7666 1166 l 7682 1186 l 7697 1206 l 7711 1226 l 7724 1246 l
 7736 1265 l 7747 1283 l 7758 1302 l 7768 1320 l 7779 1343 l
 7790 1366 l 7801 1390 l 7810 1413 l 7819 1437 l 7828 1462 l
 7836 1486 l 7843 1511 l 7849 1537 l 7855 1562 l 7860 1588 l
 7864 1613 l 7868 1638 l 7870 1663 l 7873 1687 l 7874 1711 l
 7875 1734 l 7876 1757 l 7875 1779 l 7875 1800 l 7874 1822 l
 7873 1843 l 7871 1866 l 7868 1889 l 7865 1912 l 7862 1936 l
 7858 1961 l 7853 1986 l 7848 2011 l 7842 2036 l 7836 2062 l
 7829 2087 l 7821 2112 l 7814 2137 l 7805 2161 l 7797 2184 l
 7788 2207 l 7779 2228 l 7770 2248 l 7760 2268 l 7750 2287 l
 7740 2305 l 7729 2323 l 7718 2340 l 7707 2357 l 7694 2374 l
 7681 2392 l 7668 2409 l 7653 2426 l 7638 2442 l 7622 2459 l
 7606 2475 l 7590 2491 l 7573 2506 l 7556 2521 l 7539 2535 l
 7523 2549 l 7506 2562 l 7489 2574 l 7473 2586 l 7456 2597 l
 7440 2608 l 7423 2618 l 7406 2628 l 7389 2639 l 7370 2649 l
 7351 2659 l 7332 2669 l 7311 2679 l 7290 2688 l 7268 2698 l
 7246 2706 l 7223 2715 l 7200 2722 l 7177 2729 l 7153 2736 l
 7130 2742 l 7107 2747 l 7084 2751 l 7061 2755 l 7038 2758 l
 7015 2760 l 6994 2762 l 6972 2763 l 6949 2763 l 6925 2763 l
 6901 2763 l 6876 2762 l 6850 2760 l 6824 2758 l 6797 2755 l
 6770 2751 l 6742 2747 l 6715 2742 l 6688 2737 l 6661 2731 l
 6635 2725 l 6609 2718 l 6584 2710 l 6560 2703 l 6536 2694 l
 6513 2686 l 6491 2677 l 6470 2668 l 6449 2658 l 6428 2647 l
 6407 2636 l 6387 2624 l 6366 2611 l 6345 2597 l 6324 2583 l
 6304 2567 l 6283 2551 l 6263 2535 l 6243 2517 l 6224 2500 l
 6205 2482 l 6187 2463 l 6170 2445 l 6154 2426 l 6138 2408 l
 6124 2389 l 6110 2371 l 6097 2352 l 6084 2334 l 6073 2315 l
 6061 2296 l 6050 2277 l 6040 2257 l 6029 2236 l 6019 2215 l
 6010 2193 l 6000 2171 l 5991 2147 l 5983 2124 l 5975 2100 l
 5967 2075 l 5960 2051 l 5954 2027 l 5948 2003 l 5943 1979 l
 5939 1955 l 5935 1932 l 5932 1910 l 5929 1888 l 5927 1867 l
 5926 1846 l 5925 1825 l 5925 1802 l 5925 1780 l 5926 1757 l
 5927 1733 l 5929 1709 l 5932 1685 l 5936 1660 l 5941 1636 l
 5946 1611 l 5951 1586 l 5958 1561 l 5965 1536 l 5973 1512 l
 5981 1489 l 5989 1466 l 5998 1444 l 6007 1423 l 6017 1402 l
 6027 1382 l 6038 1363 l 6047 1345 l 6058 1327 l 6069 1309 l
 6081 1291 l 6094 1271 l 6108 1251 l 6124 1230 l 6141 1208 l
 6160 1184 l 6180 1158 l 6201 1132 l 6223 1105 l 6245 1078 l
 6267 1052 l 6287 1028 l 6305 1007 l 6320 989 l 6345 960 l gs col0 s gr gr

% arrowhead
n 6094 1159 m 6327 980 l 6185 1238 l 6171 1162 l 6094 1159 l  cp gs 0.00 setgray ef gr  col0 s
% Polyline
gs  clippath
2598 2765 m 2305 2796 l 2561 2651 l 2247 2753 l 2284 2867 l cp
clip
n 3780 4185 m 3782 4182 l 3788 4176 l 3797 4166 l 3811 4150 l 3828 4131 l
 3849 4107 l 3873 4082 l 3897 4056 l 3922 4029 l 3946 4004 l
 3969 3981 l 3991 3959 l 4012 3940 l 4032 3922 l 4051 3906 l
 4069 3892 l 4087 3878 l 4104 3866 l 4123 3855 l 4141 3844 l
 4160 3834 l 4180 3824 l 4200 3815 l 4221 3806 l 4243 3798 l
 4266 3790 l 4290 3783 l 4314 3776 l 4338 3770 l 4363 3765 l
 4387 3760 l 4412 3756 l 4436 3753 l 4459 3751 l 4482 3749 l
 4505 3749 l 4527 3748 l 4549 3749 l 4570 3750 l 4591 3752 l
 4613 3754 l 4634 3757 l 4657 3761 l 4679 3765 l 4702 3771 l
 4725 3777 l 4748 3784 l 4771 3791 l 4794 3800 l 4817 3809 l
 4839 3819 l 4860 3830 l 4881 3840 l 4901 3852 l 4920 3864 l
 4938 3876 l 4955 3889 l 4972 3902 l 4988 3915 l 5003 3929 l
 5018 3944 l 5032 3960 l 5047 3976 l 5061 3994 l 5075 4012 l
 5088 4032 l 5101 4052 l 5114 4073 l 5126 4094 l 5138 4116 l
 5149 4138 l 5159 4160 l 5168 4182 l 5177 4204 l 5184 4225 l
 5191 4246 l 5197 4267 l 5203 4287 l 5208 4308 l 5212 4328 l
 5215 4348 l 5218 4369 l 5221 4391 l 5223 4413 l 5224 4436 l
 5224 4459 l 5224 4482 l 5223 4506 l 5221 4530 l 5219 4554 l
 5215 4578 l 5211 4601 l 5206 4624 l 5200 4647 l 5194 4669 l
 5187 4691 l 5180 4712 l 5171 4732 l 5163 4753 l 5154 4771 l
 5144 4789 l 5134 4808 l 5123 4827 l 5111 4846 l 5098 4866 l
 5084 4885 l 5069 4905 l 5053 4925 l 5037 4944 l 5020 4963 l
 5002 4982 l 4984 5000 l 4965 5017 l 4946 5034 l 4927 5050 l
 4908 5065 l 4888 5078 l 4869 5091 l 4849 5103 l 4830 5115 l
 4810 5125 l 4791 5134 l 4772 5142 l 4753 5150 l 4732 5157 l
 4711 5164 l 4689 5171 l 4665 5177 l 4641 5182 l 4616 5187 l
 4589 5191 l 4562 5195 l 4534 5199 l 4505 5201 l 4476 5204 l
 4446 5205 l 4416 5207 l 4385 5207 l 4354 5207 l 4323 5207 l
 4292 5206 l 4260 5205 l 4228 5204 l 4196 5202 l 4163 5200 l
 4140 5198 l 4117 5197 l 4093 5195 l 4069 5192 l 4044 5189 l
 4018 5186 l 3991 5183 l 3964 5178 l 3935 5174 l 3906 5168 l
 3876 5162 l 3844 5155 l 3812 5148 l 3779 5139 l 3745 5130 l
 3711 5120 l 3675 5108 l 3639 5096 l 3603 5083 l 3565 5069 l
 3528 5053 l 3489 5037 l 3451 5020 l 3412 5001 l 3373 4982 l
 3333 4962 l 3294 4940 l 3254 4918 l 3214 4894 l 3175 4870 l
 3134 4844 l 3094 4817 l 3054 4789 l 3013 4760 l 2972 4729 l
 2930 4698 l 2898 4673 l 2866 4647 l 2834 4620 l 2801 4592 l
 2767 4564 l 2733 4534 l 2698 4503 l 2663 4472 l 2628 4439 l
 2592 4406 l 2555 4371 l 2518 4335 l 2480 4299 l 2443 4262 l
 2404 4224 l 2366 4185 l 2327 4145 l 2288 4104 l 2249 4063 l
 2210 4022 l 2171 3980 l 2132 3937 l 2093 3894 l 2054 3851 l
 2016 3808 l 1978 3764 l 1940 3721 l 1903 3677 l 1866 3634 l
 1830 3591 l 1795 3548 l 1761 3505 l 1727 3463 l 1694 3422 l
 1662 3381 l 1631 3340 l 1600 3300 l 1571 3261 l 1543 3222 l
 1515 3184 l 1489 3147 l 1463 3110 l 1439 3074 l 1415 3039 l
 1392 3005 l 1371 2971 l 1350 2938 l 1330 2905 l 1303 2860 l
 1278 2816 l 1255 2772 l 1232 2729 l 1212 2687 l 1192 2645 l
 1174 2604 l 1157 2563 l 1141 2522 l 1127 2482 l 1114 2442 l
 1102 2403 l 1091 2364 l 1081 2326 l 1073 2289 l 1066 2252 l
 1059 2216 l 1054 2181 l 1050 2147 l 1047 2113 l 1045 2081 l
 1044 2050 l 1043 2019 l 1043 1990 l 1044 1961 l 1046 1934 l
 1048 1907 l 1050 1881 l 1053 1857 l 1057 1833 l 1060 1809 l
 1064 1787 l 1068 1764 l 1073 1743 l 1079 1712 l 1085 1682 l
 1092 1652 l 1099 1622 l 1106 1593 l 1114 1563 l 1123 1533 l
 1132 1504 l 1141 1474 l 1151 1445 l 1161 1416 l 1172 1388 l
 1183 1360 l 1195 1333 l 1207 1307 l 1219 1282 l 1231 1258 l
 1243 1235 l 1256 1213 l 1269 1191 l 1282 1171 l 1295 1152 l
 1309 1133 l 1323 1115 l 1338 1096 l 1354 1077 l 1371 1059 l
 1389 1041 l 1408 1023 l 1428 1005 l 1449 988 l 1470 971 l
 1492 954 l 1515 938 l 1539 922 l 1563 908 l 1587 894 l
 1612 881 l 1636 868 l 1660 857 l 1685 847 l 1709 837 l
 1732 829 l 1756 821 l 1779 814 l 1803 808 l 1826 802 l
 1850 797 l 1875 792 l 1900 789 l 1925 785 l 1952 783 l
 1979 780 l 2007 779 l 2035 778 l 2063 778 l 2092 779 l
 2120 780 l 2149 781 l 2177 784 l 2204 787 l 2231 790 l
 2257 794 l 2282 798 l 2307 803 l 2331 808 l 2355 814 l
 2378 820 l 2400 826 l 2423 833 l 2446 841 l 2469 849 l
 2492 858 l 2515 868 l 2538 878 l 2562 890 l 2585 901 l
 2609 914 l 2632 927 l 2655 941 l 2677 955 l 2699 970 l
 2720 985 l 2740 1000 l 2759 1016 l 2778 1032 l 2796 1048 l
 2813 1064 l 2829 1081 l 2845 1098 l 2860 1115 l 2876 1133 l
 2891 1152 l 2905 1172 l 2920 1193 l 2934 1215 l 2949 1237 l
 2963 1261 l 2976 1285 l 2989 1310 l 3002 1336 l 3014 1362 l
 3025 1388 l 3036 1414 l 3045 1441 l 3054 1467 l 3062 1493 l
 3069 1518 l 3076 1543 l 3081 1568 l 3086 1593 l 3090 1618 l
 3093 1640 l 3096 1663 l 3098 1687 l 3099 1710 l 3100 1735 l
 3101 1760 l 3101 1785 l 3100 1811 l 3098 1837 l 3096 1864 l
 3094 1891 l 3090 1918 l 3087 1944 l 3082 1971 l 3077 1997 l
 3072 2023 l 3066 2048 l 3060 2072 l 3053 2096 l 3046 2119 l
 3039 2141 l 3031 2163 l 3023 2184 l 3015 2205 l 3005 2227 l
 2995 2250 l 2984 2272 l 2973 2294 l 2960 2316 l 2947 2339 l
 2934 2361 l 2919 2383 l 2904 2405 l 2888 2427 l 2872 2448 l
 2856 2469 l 2839 2489 l 2822 2508 l 2805 2527 l 2789 2544 l
 2772 2561 l 2755 2576 l 2739 2591 l 2722 2605 l 2706 2618 l
 2690 2630 l 2672 2643 l 2654 2655 l 2635 2666 l 2615 2677 l
 2594 2688 l 2571 2699 l 2546 2710 l 2520 2721 l 2491 2732 l
 2461 2743 l 2430 2754 l 2399 2765 l 2370 2776 l 2343 2785 l
 2320 2792 l 2280 2805 l gs col0 s gr gr

% arrowhead
n 2598 2765 m 2305 2796 l 2561 2651 l 2534 2722 l 2598 2765 l  cp gs 0.00 setgray ef gr  col0 s
% Polyline
gs  clippath
5382 3862 m 5444 4149 l 5273 3911 l 5407 4212 l 5517 4163 l cp
clip
n 6671 2537 m 6668 2535 l 6662 2530 l 6650 2522 l 6633 2510 l 6612 2495 l
 6587 2476 l 6559 2456 l 6530 2434 l 6501 2412 l 6474 2391 l
 6448 2370 l 6424 2350 l 6403 2332 l 6383 2314 l 6365 2297 l
 6349 2281 l 6334 2264 l 6320 2248 l 6307 2231 l 6294 2214 l
 6282 2196 l 6271 2177 l 6259 2158 l 6248 2138 l 6238 2117 l
 6227 2095 l 6218 2072 l 6209 2049 l 6200 2026 l 6192 2002 l
 6185 1978 l 6179 1954 l 6173 1930 l 6169 1907 l 6165 1884 l
 6162 1862 l 6159 1840 l 6157 1819 l 6156 1797 l 6156 1776 l
 6156 1754 l 6156 1732 l 6158 1710 l 6160 1687 l 6163 1664 l
 6167 1640 l 6171 1617 l 6176 1593 l 6183 1569 l 6189 1546 l
 6197 1522 l 6205 1500 l 6214 1478 l 6223 1457 l 6233 1437 l
 6243 1418 l 6254 1399 l 6265 1382 l 6277 1365 l 6289 1348 l
 6302 1332 l 6316 1316 l 6331 1299 l 6347 1284 l 6364 1268 l
 6382 1252 l 6401 1237 l 6420 1222 l 6440 1208 l 6461 1194 l
 6481 1181 l 6502 1169 l 6523 1157 l 6544 1146 l 6564 1137 l
 6585 1128 l 6605 1119 l 6624 1112 l 6644 1105 l 6664 1099 l
 6684 1093 l 6704 1088 l 6726 1083 l 6747 1079 l 6770 1075 l
 6793 1073 l 6816 1070 l 6840 1069 l 6864 1068 l 6888 1068 l
 6912 1069 l 6936 1071 l 6959 1073 l 6983 1077 l 7005 1081 l
 7027 1085 l 7049 1091 l 7070 1097 l 7091 1103 l 7111 1110 l
 7130 1118 l 7150 1126 l 7170 1135 l 7190 1145 l 7211 1156 l
 7232 1168 l 7253 1180 l 7274 1194 l 7295 1208 l 7316 1223 l
 7336 1239 l 7356 1255 l 7375 1272 l 7394 1289 l 7411 1306 l
 7428 1324 l 7444 1342 l 7459 1360 l 7473 1378 l 7486 1396 l
 7498 1415 l 7509 1432 l 7519 1450 l 7529 1469 l 7538 1489 l
 7548 1509 l 7556 1531 l 7565 1553 l 7573 1577 l 7580 1601 l
 7587 1627 l 7594 1654 l 7600 1681 l 7606 1710 l 7611 1738 l
 7616 1768 l 7620 1798 l 7624 1828 l 7628 1859 l 7631 1890 l
 7633 1921 l 7636 1953 l 7638 1985 l 7639 2018 l 7641 2051 l
 7641 2073 l 7642 2096 l 7642 2120 l 7643 2145 l 7642 2170 l
 7642 2196 l 7641 2223 l 7640 2251 l 7638 2279 l 7636 2309 l
 7633 2340 l 7630 2372 l 7625 2404 l 7620 2438 l 7615 2473 l
 7608 2508 l 7600 2545 l 7592 2582 l 7583 2620 l 7572 2658 l
 7561 2698 l 7549 2737 l 7536 2777 l 7522 2818 l 7507 2859 l
 7490 2900 l 7473 2942 l 7455 2984 l 7436 3026 l 7416 3068 l
 7394 3110 l 7372 3153 l 7348 3196 l 7323 3240 l 7297 3284 l
 7270 3329 l 7249 3363 l 7226 3398 l 7203 3433 l 7179 3469 l
 7154 3505 l 7128 3542 l 7101 3580 l 7073 3618 l 7045 3657 l
 7015 3696 l 6985 3736 l 6953 3777 l 6921 3818 l 6888 3860 l
 6854 3902 l 6819 3944 l 6784 3987 l 6748 4030 l 6711 4074 l
 6674 4117 l 6637 4161 l 6599 4204 l 6560 4247 l 6522 4291 l
 6483 4334 l 6444 4376 l 6405 4419 l 6366 4461 l 6327 4502 l
 6288 4543 l 6250 4583 l 6212 4622 l 6174 4660 l 6136 4698 l
 6099 4735 l 6063 4770 l 6027 4805 l 5991 4839 l 5956 4872 l
 5922 4904 l 5888 4935 l 5855 4965 l 5823 4994 l 5791 5022 l
 5760 5048 l 5730 5074 l 5700 5099 l 5670 5124 l 5629 5156 l
 5589 5187 l 5550 5217 l 5511 5245 l 5473 5271 l 5435 5297 l
 5397 5321 l 5360 5344 l 5323 5366 l 5286 5386 l 5250 5406 l
 5214 5424 l 5179 5440 l 5144 5456 l 5109 5471 l 5076 5484 l
 5042 5496 l 5010 5507 l 4978 5516 l 4946 5525 l 4916 5533 l
 4886 5539 l 4857 5545 l 4829 5549 l 4802 5553 l 4775 5556 l
 4749 5558 l 4724 5560 l 4700 5561 l 4676 5561 l 4653 5561 l
 4630 5561 l 4607 5560 l 4585 5559 l 4556 5557 l 4527 5555 l
 4498 5553 l 4469 5550 l 4439 5546 l 4409 5542 l 4379 5537 l
 4349 5532 l 4319 5526 l 4288 5520 l 4258 5513 l 4228 5506 l
 4198 5498 l 4168 5489 l 4139 5480 l 4111 5471 l 4084 5461 l
 4057 5451 l 4031 5440 l 4007 5429 l 3983 5418 l 3960 5407 l
 3938 5395 l 3917 5383 l 3897 5371 l 3877 5359 l 3857 5344 l
 3837 5329 l 3817 5314 l 3797 5298 l 3778 5280 l 3759 5262 l
 3741 5244 l 3722 5224 l 3704 5203 l 3687 5182 l 3670 5161 l
 3654 5138 l 3638 5116 l 3623 5093 l 3609 5070 l 3595 5047 l
 3583 5024 l 3571 5001 l 3560 4978 l 3550 4956 l 3540 4933 l
 3531 4911 l 3523 4889 l 3516 4868 l 3509 4845 l 3502 4823 l
 3496 4800 l 3490 4777 l 3485 4753 l 3480 4729 l 3475 4704 l
 3471 4679 l 3468 4653 l 3464 4627 l 3462 4601 l 3460 4574 l
 3458 4548 l 3457 4521 l 3456 4495 l 3456 4470 l 3456 4445 l
 3457 4420 l 3459 4396 l 3460 4373 l 3463 4350 l 3465 4328 l
 3468 4306 l 3471 4284 l 3475 4261 l 3480 4238 l 3485 4214 l
 3491 4191 l 3497 4167 l 3505 4142 l 3513 4118 l 3521 4093 l
 3531 4069 l 3541 4044 l 3551 4020 l 3563 3996 l 3575 3972 l
 3587 3949 l 3600 3926 l 3613 3905 l 3626 3884 l 3640 3864 l
 3654 3844 l 3669 3826 l 3683 3808 l 3699 3790 l 3714 3773 l
 3731 3756 l 3748 3739 l 3766 3722 l 3786 3705 l 3806 3689 l
 3827 3672 l 3849 3656 l 3871 3640 l 3895 3624 l 3919 3609 l
 3944 3595 l 3969 3581 l 3994 3568 l 4019 3555 l 4044 3544 l
 4069 3533 l 4093 3523 l 4118 3514 l 4142 3506 l 4166 3499 l
 4190 3492 l 4212 3487 l 4235 3482 l 4258 3477 l 4281 3473 l
 4305 3470 l 4330 3467 l 4355 3464 l 4381 3462 l 4408 3461 l
 4434 3460 l 4461 3460 l 4488 3460 l 4515 3461 l 4542 3463 l
 4568 3465 l 4594 3467 l 4620 3471 l 4645 3474 l 4669 3478 l
 4693 3483 l 4716 3488 l 4738 3493 l 4760 3499 l 4782 3505 l
 4805 3512 l 4828 3520 l 4851 3529 l 4875 3538 l 4898 3548 l
 4922 3558 l 4945 3570 l 4969 3582 l 4992 3594 l 5016 3608 l
 5039 3621 l 5061 3636 l 5083 3650 l 5104 3665 l 5124 3680 l
 5143 3695 l 5161 3710 l 5178 3724 l 5195 3739 l 5210 3754 l
 5225 3769 l 5239 3784 l 5253 3800 l 5267 3817 l 5280 3835 l
 5293 3854 l 5306 3874 l 5320 3895 l 5333 3919 l 5347 3944 l
 5361 3971 l 5375 4000 l 5390 4030 l 5404 4059 l 5417 4088 l
 5429 4114 l 5439 4136 l 5456 4174 l gs col0 s gr gr

% arrowhead
n 5382 3862 m 5444 4149 l 5273 3911 l 5347 3930 l 5382 3862 l  cp gs 0.00 setgray ef gr  col0 s
/Symbol ff 450.00 scf sf
4380 4545 m
gs 1 -1 sc (L) col0 sh gr
/Symbol ff 450.00 scf sf
6765 1845 m
gs 1 -1 sc (L) col0 sh gr
/Symbol ff 450.00 scf sf
1965 1830 m
gs 1 -1 sc (L) col0 sh gr
/Times-Roman ff 225.00 scf sf
4680 4620 m
gs 1 -1 sc (1) col0 sh gr
/Times-Roman ff 225.00 scf sf
7065 1890 m
gs 1 -1 sc (3) col0 sh gr
/Times-Roman ff 225.00 scf sf
2295 1905 m
gs 1 -1 sc (2) col0 sh gr
$F2psEnd
rs
EndOfTheIncludedPostscriptMagicCookie

\closepsdump

% Here is the PostScript for homfig2.fig:
\message{Writing file homfig2.fig}
\psdump{homfig2.fig}#FIG 3.2
Landscape
Center
Inches
Letter
100.00
Single
-2
1200 2
1 3 0 4 0 7 0 0 -1 0.000 1 0.0000 2100 1800 900 900 2100 1800 3000 1800
1 3 0 4 0 7 0 0 -1 0.000 1 0.0000 6900 1800 849 849 6900 1800 7500 2400
1 3 0 4 0 7 0 0 -1 0.000 1 0.0000 4500 4500 900 900 4500 4500 4500 3600
2 1 0 1 0 7 0 0 -1 0.000 0 0 -1 1 0 2
2 1 1.00 180.00 360.00
 4830 327 4881 327
2 1 0 1 0 7 0 0 -1 0.000 0 0 -1 1 0 2
2 1 1.00 180.00 360.00
 2106 3906 2076 3873
2 1 0 1 0 7 0 0 -1 0.000 0 0 -1 1 0 2
2 1 1.00 180.00 360.00
 6690 4101 6657 4140
3 0 0 1 0 7 0 0 -1 0.000 0 1 0 21
2 1 1.00 120.00 240.00
 2794 1510 2836 1998 2571 2411 2136 2613 1669 2464 1372 2125
 1330 1552 1712 1107 3270 300 5745 300 7320 735 7845 1335
 7905 1845 7770 2340 7455 2625 7050 2805 6435 2715 6030 2340
 5880 1815 6000 1350 6345 960
 0.000 1.000 1.000 1.000 1.000 1.000 1.000 1.000
 1.000 1.000 1.000 1.000 1.000 1.000 1.000 1.000
 1.000 1.000 1.000 1.000 0.000
3 0 0 1 0 7 0 0 -1 0.000 0 1 0 19
2 1 1.00 120.00 240.00
 3780 4185 4095 3810 4575 3705 5025 3870 5250 4305 5220 4755
 4845 5190 4260 5235 3090 5070 960 2670 1050 1680 1275 1065
 1785 750 2400 780 2880 1050 3150 1605 3060 2235 2700 2685
 2280 2805
 0.000 1.000 1.000 1.000 1.000 1.000 1.000 1.000
 1.000 1.000 1.000 1.000 1.000 1.000 1.000 1.000
 1.000 1.000 0.000
3 0 0 1 0 7 0 0 -1 0.000 0 1 0 19
2 1 1.00 120.00 240.00
 6671 2537 6265 2263 6111 1797 6228 1332 6637 1063 7088 1046
 7559 1373 7665 1950 7624 3131 5459 5500 4560 5610 3810 5415
 3464 4881 3429 4266 3648 3760 4171 3434 4807 3457 5292 3768
 5456 4174
 0.000 1.000 1.000 1.000 1.000 1.000 1.000 1.000
 1.000 1.000 1.000 1.000 1.000 1.000 1.000 1.000
 1.000 1.000 0.000
4 0 0 0 0 32 30 0.0000 4 300 300 4380 4545 L\001
4 0 0 0 0 32 30 0.0000 4 300 300 6765 1845 L\001
4 0 0 0 0 32 30 0.0000 4 300 300 1965 1830 L\001
4 0 0 0 0 1 15 0.0000 0 150 105 4680 4620 1\001
4 0 0 0 0 1 15 0.0000 0 150 105 7065 1890 3\001
4 0 0 0 0 1 15 0.0000 0 150 105 2295 1905 2\001
EndOfTheIncludedPostscriptMagicCookie

\closepsdump

% Finally, here is the rest of generic.tex:
% \documentstyle[12pt,amssymb]{amsart}
%\usepackage{amssymb}
%\documentstyle[11pt]{article}
\include{epsf}
\topmargin0in
\textheight8.5in
\oddsidemargin0.2in
\evensidemargin0.2in
\textwidth6in
\advance\hoffset by -0.5 truecm

\newtheorem{Theorem}{Theorem}[section]
\newtheorem{Lemma}[Theorem]{Lemma}
\newtheorem{Corollary}[Theorem]{Corollary}

\def\df{\buildrel \rm def \over =}
\newtheorem{Proposition}[Theorem]{Proposition}
\newtheorem{Conjecture}[Theorem]{Conjecture}
%\newtheorem{Definition}[Theorem]{Definition}
%\newtheorem{Example}[Theorem]{Example}
%\newtheorem{Observation}[Theorem]{Observation}
\newtheorem{Remark}[Theorem]{Remark}
\def \rk{{\mbox {rk}}\,}
\def \dim{{\mbox {dim}}\,}
\def\df{\buildrel \rm def \over =}
\def\PROOF{{\em Proof.}\;}
\def\QED{~\hfill~ $\diamond$ \vspace{7mm}}
\def\ind{{\mbox {ind}}\,}
\def\Vol{\mbox{Vol}}
\def \a{\alpha}
\def \be{\beta}
\def \ga{\gamma}
\def \Ga{\Gamma}
\def \de{\delta}
\def \e{\varepsilon}
\def \la{\lambda}
\def \La{\Lambda}
\def \vr{\varphi}
\def \om{\omega}
\def \Om{\Omega}
\def \nulo{\text{\o}}
\def \re{{\Bbb R}}
\def \na{{\Bbb N}}
\def \lv{\left\vert}
\def \rv{\right\vert}
\def \lV{\left\Vert}
\def \rV{\right\Vert}
\def \ov{\overline}
\def \ul{\underline}
\def \then{\Longrightarrow}
\def \lqqd{\quad\blacksquare}
\def \dsize{\displaystyle}
\def \tsize{\textstyle}
\def \ssize{\scriptstyle}
\def \sssize{\scriptscriptstyle}
\def\V{\bf V}
\def\W{\bf W}
\def\p{\Phi_{k}}
\def\q{\widehat{\Phi}_{k}}
\def\qc{\widehat{\Phi}_{c}}
\def\r{\widetilde{\Phi}_{k}}
\def\s{\overline{\Phi}_{k}}
\def\w{W_{k}(x,r)}
\def\f{\tilde{f}_{k}}
\def \e{\varepsilon}
\def \vr{\varphi}
\def \lv{\left\vert}
\def \rv{\right\vert}
\def \ov{\overline}
\def \tx{{\widehat{x}}}
\def \ze{{\Bbb Z}}
\def \ra{{\Bbb Q}}
\def \hS{{\widehat{\Sigma}}}
\def \cG{{\mathcal G}}
 \def \cU{{\mathcal U}}
 \def \cV{{\mathcal V}}
 \def \M{{\mathcal M}}
 \def \S{\Sigma}
\def \supp{\mbox{\rm supp}}


\begin{document}
\begin{thebibliography}{aa}


\bibitem{Bo1} S. Bolotin, {\it Invariant sets of Hamiltonian systems and variational methods,} Proceedings of the International Congress of Mathematicians, Vol. 1,2 (Z\"urich, 1994) 1169--1178, Birkh\"auser, Basel, 1995.

\bibitem{Bo2} S. Bolotin, {\it Homoclinic orbits in invariant tori of Hamiltonian systems,} Dynamical systems in classical mechanics, V.V. Kozlov ed, Amer. Math. Soc. Transl. Ser. 2, {\bf 168}, Amer. Math. Soc., Providence, RI, 1995, 21--90.

\bibitem{Bo3} S. Bolotin, {\it Homoclinic trajectories of invariant sets
of Hamiltonian systems,} Nonlinear Differ. Equ. Appl. {\bf 4}, N.3, (1997) 
359--389.

\bibitem{CDI} G. Contreras, J. Delgado, R. Iturriaga, {\it Lagrangian flows: the
 dynamics of globally minimizing orbits II}, Bol. Soc. Bras. Mat.
Vol. {\bf 28}, N.2, (1997) 155-196. 

\bibitem{CDII} G. Contreras, R. Iturriaga, {\it Convex Hamiltonians
without conjugate points}, to appear in Ergod. Th and Dynam. Sys., preprint available via internet at ``http://www.ma.utexas.edu/mp\_arc".

%\bibitem{CI} G. Contreras, R. Iturriaga, {\it Global minimizers of autonomous
%Lagrangians}, 22 Coloquio Bras. de Mat. IMPA. R\'\i o de Janeiro, 1999. 
 
\bibitem{Fa1} A. Fathi, {\it Th\'eor\`eme KAM faible et Th\'eorie de
Mather sur les systems Lagrangiens}, C.R. Acad. Sci. Paris, t. 324, S\'{e}rie I (1997) 1043--1046.

\bibitem{Fa2} A. Fathi, {\it Solutions KAM faibles conjugu\'ees
 et barri\`eres de Peierls}, C.R. Acad. Sci. Paris, t. 325, S\'{e}rie I
 (1997) 649--652. 

\bibitem{Fa3} A. Fathi, {\it Orbites h\'et\'eroclines et ensemble
de Peierls}, C. R. Acad. Sci. Paris, t. 326, S\'{e}rie I
(1998) 1213--1216.

\bibitem{Fa4} A. Fathi, {\it Dynamique Lagrangiene}, Lecture Notes, ENS Lyon, 1998.

\bibitem{Ma} R. Ma\~n\'e, {\it Lagrangian flows: the dynamics
 of globally minimizing orbits},
 International Congress on Dynamical Systems in Montevideo
 (a tribute to Ricardo Ma\~n\'e), F. Ledrappier, J. Lewowicz,
 S. Newhouse eds, Pitman Research Notes in Math. {\bf 362} (1996) 120--131.
Reprinted in Bol. Soc. Bras. Mat. Vol~{\bf
                   28}, N. 2, (1997) 141-153.

\bibitem{Ma1} R. Ma\~n\'e, {\it Generic properties and problems
of minimizing measures of Lagrangian systems}, Nonlinearity {\bf 9}
 (1996) 273--310. 

\bibitem{M} J. Mather, {\it Action minimizing measures for positive
 definite Lagrangian systems}, Math. Z. {\bf 207} (1991) 169--207.

\bibitem{M'} J. Mather, {\it Variational construction of connecting orbits}, 
Ann. Inst. Fourier {\bf 43} (1993) 1349--1386.


\end{thebibliography}

\title[Connecting orbits]{Connecting orbits between static
classes for generic Lagrangian systems}

\author[G. Contreras]{Gonzalo Contreras}\thanks{Gonzalo Contreras was
partially supported by CNPq-Brasil and CONACYT-M\'exico.}
\address{CIMAT \\
         A.P. 402, 3600 \\
         Guanajuato. Gto. \\
         M\'exico.}
\email{gonzalo@@cimat.mx}


\author[G. P. Paternain]{Gabriel P. Paternain}
\address{Centro de Matem\'atica\\
Facultad de Ciencias\\
Igu\'a 4225\\
11400 Montevideo\\
Uruguay}
\email {gabriel@@cmat.edu.uy}
\curraddr{CIMAT \\
         A.P. 402, 3600 \\
         Guanajuato. Gto. \\
         M\'exico.}
\email{paternain@@cimat.mx}

\subjclass{58F05, 58F11}



\date{1999}
%\maketitle
\begin{abstract} Let $L$ be a $C^{\infty}$ convex superlinear Lagrangian
on a closed manifold $M$. We show that if the number
of static classes is finite, then
there exist chains of semistatic orbits that connect
 any two given static classes.
Using this property we show that if there is only
one static class, then
the homoclinic orbits to the set of static orbits
 generate over $\re$ the relative homology of the pair
$(M,U)$, where $U$ is a sufficiently small connected 
neighborhood of the set of static orbits in $M$.

We show that generically in the sense of Ma\~n\'e \cite{Ma} the set
of semistatic orbits coincides with the support of a uniquely minimizing
 measure, therefore generically, the homoclinic orbits to the support
 of the minimizing measure
 generate over $\re$ the relative homology of the pair
$(M,U)$, where $U$ is a sufficiently small connected
neighborhood of the projection
of the support of the measure to $M$.
 
This last result was obtained -with a different proof- by S. Bolotin in \cite{Bo1,Bo2} assuming the existence of a $C^{1+Lip}$ function
$f:M\to \re$ such that $L+c-df\geq 0$, where $c$ is the critical value
of $L$.


Finally we obtain two consequences.
The first one says that if $M$ is a closed
 manifold with first Betti
number $\geq 2$ then there exists a generic set ${\cal O}\subset C^{\infty}(M,\re)$
such that if $\psi\in{\cal O}$ the Lagrangian $L+\psi$ has a unique
minimizing measure and this measure
is uniquely ergodic. When this measure is supported on a periodic orbit, this
orbit is hyperbolic and the stable and unstable manifolds have transverse homoclinic intersections. 
The second consequence says that if $M$ is a closed manifold with
first Betti number different from zero and if
$L$ is a symmetric Lagrangian, then there exists a generic
set ${\cal O}\subset C^{\infty}(M,\re)$
such that if $\psi\in{\cal O}$, then $L+\psi$ has a unique
minimizing measure and this measure
is supported on a hyperbolic fixed point
whose stable and unstable manifolds have transverse
homoclinic intersections. 

\end{abstract}

\maketitle

\section{Introduction}

Let $M$ be a closed connected smooth manifold and let $L:TM\to \re$
be a smooth convex superlinear Lagrangian. This means that $L$ restricted
to each $T_{x}M$ has positive definite Hessian and that for some Riemannian
metric we have that
\[\lim_{|v|\to \infty}\frac{L(x,v)}{|v|}=\infty,\]
 uniformly on $x\in M.$
Since $M$ is compact, the extremals of $L$ give rise to a complete flow
$f_{t}:TM\to TM$ called the Euler-Lagrange flow of the Lagrangian.
The extremals are solutions of the Euler-Lagrange equation
which in local coordinates is given by

$$\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\frac{d}{dt}\frac{\partial L}{\partial v}=\frac{\partial L}{\partial x}.
\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\mbox{(E-L)}$$
The {\it energy} $E:TM\rightarrow \re$ is defined by
\[E(x,v)=\frac{\partial L}{\partial v}(x,v).v-L(x,v).\]
Since $L$ is autonomous, $E$ is a first integral of the flow $f_{t}$.

Recall that the action of the Lagrangian $L$ on
 an absolutely continuous curve $\ga:[a,b]\rightarrow M$ is defined by
\[A_{L}(\ga)=\int_{a}^{b}L(\ga(t),\dot{\ga}(t))\,dt.\]
Given two points, $x_{1}$ and $x_{2}$ in $M$ and $T>0$
 denote by ${\cal C}_{T}(x_{1},x_{2})$ the set of absolutely continuous curves $\ga:[0,T]\rightarrow M$, with $\ga(0)=x_{1}$ and $\ga(T)=x_{2}$.
For each $k\in \re$ we define the {\it action potential} $\Phi_{k}:M\times M\rightarrow \re$ by
\[\Phi_{k}(x_{1},x_{2})=\inf\{A_{L+k}(\ga):\;\ga\in \cup_{T>0}{\cal C}_{T}(x_{1},x_{2})\}.\]

The {\it critical value of $L$}, which was introduced
 by Ma\~n\'e in \cite{Ma}, is the real number $c(L)$ defined as
the infimum of $k\in\re$ such that for some $x\in M$, $\Phi_{k}(x,x)>-\infty$.
Since $L$ is convex and superlinear and $M$ is compact such a number exists
 and it has various important properties that we review in Section 2.
We briefly mention a few of them since we shall need them below.
For any $k\geq c(L)$, the action potential $\Phi_{k}$ is a Lipschitz
function that satisfies a triangle inequality. In general the action potential
is not symmetric but if we define $d_{k}:M\times M\to\re$ by setting
\[d_{k}(x,y)=\Phi_{k}(x,y)+\Phi_{k}(y,x),\]
then $d_{k}$ is a distance function for all $k>c(L)$ and a pseudo-distance
for $k=c(L)$.

Since $d_{k}\geq 0$, for every absolutely continuous curve $\ga:[a,b]\to M$
and all $k\geq c(L)$ we have
\begin{equation}
A_{L+k}(\ga)\geq \Phi_{k}(\ga(a),\ga(b))\geq -\Phi_{k}(\ga(b),\ga(a)).\label{se}
\end{equation}
Set $c=c(L)$. We say that an absolutely continuous curve
 $\ga:[a,b]\to M$ is {\it semistatic} if
\[A_{L+c}(\ga|_{[t_{0},t_{1}]})=\Phi_{c}(\ga(t_{0}),\ga(t_{1})),\]
for all $a<t_{0}\leq t_{1}<b$; and that it is {\it static}
if 
\[A_{L+c}(\ga|_{[t_{0},t_{1}]})=-\Phi_{c}(\ga(t_{1}),\ga(t_{0})),\]
for all $a<t_{0}\leq t_{1}<b$.
Clearly, by (\ref{se}) a static curve is semistatic.
One could also say that $\ga|_{[a,b]}$ is static
if it is semistatic and for all $a<t_{0}\leq t_{1}<b$,
 $d_{c}(\ga(t_0),\ga(t_1))=0$.
Semistatic curves are solutions of the Euler-Lagrange equation because
of their minimizing properties. Also it is not hard
to check that semistatic curves have energy precisely $c$ \cite{Ma, CDI}.
The notions of semistatic and static curves are closely related to Mather's notions
of {\it c-minimal trajectories} and {\it regular c-minimal trajectories} respectively, see \cite{M'}.

Given a vector $v\in TM$ we shall denote by $x_v:\re\to M$ the
solution of the Euler-Lagrange equation with $\dot{x}(0)=v$.

The set of vectors $v$ in $TM$ that give rise to static
 curves $x_v:\re\to M$ is a closed invariant set
 that we shall denote by $\hS:=\hS(L)$.
Similarly, 
the set of vectors $v$ in $TM$ that give rise to semistatic
 curves $x_v:\re\to M$ is a closed invariant set that
 we shall denote by $\Sigma:=\Sigma(L)$. The set $\hS$ is chain recurrent
 and the set $\Sigma$ is chain transitive \cite[Theorem V]{Ma,CDI}.
As we mentioned before $\hS\subset \Sigma$.
We need to recall (cf. Section 3, Theorem \ref{GraphProperty}) the following important {\it Lipschitz graph property} which was shown in \cite{Ma, CDI} and \cite[Theorem 6.1]{M'} that generalizes the celebrated
Lipschitz Graph Theorem of Mather \cite{M}: the set $\hS$ is a Lipschitz graph, that is, if $\pi:TM\to M$
denotes the canonical projection then the map $\pi|_{\hS}:\hS\to \pi(\hS)$ is bijective with Lipschitz inverse.
Using the graph property we can define an equivalence relation
in $\hS$ by saying that two vectors $v$ and $w$ in $\hS$ are equivalent iff $d_{c}(\pi(v),\pi(w))=0$.
The equivalence relation breaks $\hS$ into classes that we shall
call {\it static classes}.
 Let $\boldsymbol{\La}$ be the set of static classes.
Define a reflexive partial order $\preceq$ in $\boldsymbol{\La}$ by
 \begin{itemize}
 \item[(a)] $\preceq$ is reflexive.
 \item[(b)] $\preceq$ is transitive.
 \item[(c)] If there is $v\in\Sigma$ with the $\alpha$-limit set $\a(v)\subseteq\La_i$ and $\om$-limit set
            $\om(v)\subseteq\La_j$, then $\La_i\preceq\La_j$.
 \end{itemize} 


\vspace{.5cm}

\noindent {\bf Theorem A.} {\it Suppose that the number
of static classes is finite.
Then given $\La_i$ and $\La_j$ in $\boldsymbol{\La}$,
we have that $\La_i\preceq\La_j$.
  }

\vspace{.5cm}

\begin{figure}

\special{psfile=homfig2.eps hoffset=70 voffset=-170
 hscale=70 vscale=50 angle=0}

\vspace{7cm} 

\caption{Connecting orbits between static classes. The three closed
curves represent the static classes and the other curves represent
semistatic orbits connecting them.}

\label{fig1}

\end{figure}
 

Theorem A could be restated by saying that if
 the cardinality of $\boldsymbol{\La}$ is finite, then given two static
classes $\La_i$ and $\La_j$ there exist classes $\La_i=\La_1,\dots,\La_n=\La_j$
and semistatic vectors $v_1,\dots,v_{n-1}\in \Sigma$ such that
for all $1\leq k\leq n-1$ we have that $\alpha(v_{k})\subseteq \La_k$
and $\omega(v_k)\subseteq\La_{k+1}$.
In other words, between two static classes there exists a chain of static classes connected by heteroclinic semistatic orbits
(cf. Figure \ref{fig1}).

Let us assume now that $\hS$ contains only {\it one} static class.
We shall see in Section 3 (cf. Proposition \ref{ConnectedClass})
 that the static
classes are always connected, thus if we assume
 that there is only
 one static class, $\hS$ must be connected.

Given $\varepsilon>0$, let $U_{\varepsilon}$ be the $\varepsilon$-neighbourhood of $\pi(\hS)$. Since $\hS$ is connected,
the open set $U_{\varepsilon}$ is connected for $\varepsilon$ sufficiently small. 
 Let $H_{1}(M,U_{\varepsilon},\re)$ denote the first
 relative singular homology group of the pair $(M,U_{\varepsilon})$ with real coefficients. 

We shall say that an orbit of $L$ is {\it homoclinic} to a closed invariant set $K\subset TM$ if
its $\alpha$ and $\omega$-limit sets are contained in $K$.


Observe that to each homoclinic orbit $x:\re\to M$
 to the set of static orbits $\hS$ we can associate
a homology class in $H_{1}(M,U_{\varepsilon},\re)$. 
Indeed, since there exists $t_{0}>0$ such that
for all $t$ with $|t|\geq t_{0}$, $x(t)\in U_{\varepsilon}$, the class of
$x|_{[-t_{0},t_{0}]}$ defines an element in $H_{1}(M,U_{\varepsilon},\re)$.
Let us denote by ${\cal H}$ the subset of $H_{1}(M,U_{\varepsilon},\re)$ given by all the classes corresponding to homoclinic orbits to $\hS$.

In Section 4 we shall show the following result.

\vspace{.5cm}

\noindent {\bf Theorem B.} {\it Suppose that $\hS$ contains
only one static class.
 Then for any $\varepsilon$ sufficiently small
the set ${\cal H}$ generates over $\re$ the relative homology
$H_{1}(M,U_{\varepsilon},\re)$.
In particular, there exist at least ${\rm \dim}\,H_{1}(M,U_{\varepsilon},\re)$
 homoclinic orbits to the set of static orbits $\hS$.
  }

\vspace{.5cm}
 





In \cite{Ma1}, Ma\~n\'e introduced the concept of {\it generic property
of a Lagrangian }. A property $P$ is said to be {\it generic} for the
Lagrangian $L$ if there exists a generic set ${\cal O}$ (in the Baire sense) of the set $C^{\infty}(M,\re)$ of all $C^{\infty}$ functions from $M$ to $\re$
such that if $\psi\in{\cal O}$ the Lagrangian $L+\psi$ has the property $P$.
One of Ma\~n\'e's objectives in \cite{Ma1} was to show that Mather's
theory of minimizing measures becomes much more accurate and stronger
if one searches for generic properties.

Our next result describes a generic property of Lagrangians
on closed manifolds.
Let ${\cal M}(L)$ be the set of probabilities on the Borel $\sigma$-algebra of $TM$
 that have compact support and are invariant under the flow $f_{t}$.
We shall say that a measure $\mu\in {\cal M}(L)$ is {\it minimizing}
if
\[\int L\,d\mu=-c.\]
We shall denote by ${\cal M}^{0}(L)$ the set of minimizing measures. We say that a measure $\mu$ is {\it uniquely minimizing}
if the set ${\cal M}^{0}(L)$ contains $\mu$ only. 
It was shown in \cite{Ma,CDI} that a measure $\mu$ is minimizing if and only
if the support of $\mu$ is contained in $\hS$. Mather has shown in
\cite{M'} that if $\mu$ is a minimizing measure then its support
is contained in the set of Mather's regular c-minimal curves.

The following important generic property was proved in \cite{CDII} and
\cite{Ma1}.  Given a
Lagrangian $L$ there exists a generic set ${\cal O}\subset C^{\infty}(M,\re)$
such that if $\psi\in{\cal O}$ the Lagrangian $L+\psi$ has a unique
minimizing measure in ${\cal M}^{0}(L+\psi)$ and this measure
is uniquely ergodic. When this measure is supported on a periodic orbit, this
orbit is hyperbolic and {\bf if} the stable and unstable manifolds intersect, they must do it transversally. 
It is conjectured in \cite{Ma} that the unique
minimizing measure in ${\cal M}^{0}(L+\psi)$ is always supported on a periodic orbit.

We will prove in Section 5:

\vspace{.5cm}

\noindent {\bf Theorem C.} {\it  Let
    $$
    \cG_2 := \{\, \psi\in C^\infty(M,\re)\,\vert\,
    \M^{0}(L+\psi)=\{\mu\} \text{ and } \supp(\mu)=\hS(L+\psi)=\Sigma(L+\psi)\,\}.$$
Then,
\begin{itemize}
    \item[(a)] $\cG_2$ is generic in $C^\infty(M,\re)$.
    \item[(b)] If $\psi_0\in\cG_2$, then 
               $\lim_{\psi\to\psi_0}d_{H}\big(\hS(L+\psi),\,\hS(L+\psi_0)\big)=0$
               where $d_{H}$ is the Hausdorff metric between compact subsets of $TM$.
    \item[(c)] If $\psi\in C^\infty(M,\re)$, $\mu_\psi\in\M^{0}(L+\psi)$
               and $\psi_0\in\cG_2$, then
               $$
               \lim_{\psi\to\psi_0}d_{H}\big(\supp(\mu_\psi)\;,\supp(\mu_{\psi_0})\big)=0.
               $$
               \end{itemize}}


\vspace{.5cm}

Note that since $\mu$ is also uniquely ergodic, the set ${\rm supp}(\mu)$
must be a static class.
Therefore, generically, the set of static orbits contains only one
static class and it
coincides with the support of the uniquely minimizing measure.

Let us denote by $U_{\varepsilon}$ the $\varepsilon$-neighborhood
of the set ${\rm supp}(\mu)$.
From Theorems B and C we obtain right away the following
generic property.

\vspace{.5cm}

\noindent {\bf Corollary 1.} {\it Given a
Lagrangian $L$ there exists a generic set ${\cal O}\subset C^{\infty}(M,\re)$
such that if $\psi\in{\cal O}$ the Lagrangian $L+\psi$ has a unique
minimizing measure $\mu$ in ${\cal M}^{0}(L+\psi)$ and this measure
is uniquely ergodic.
For any $\varepsilon$ sufficiently small
the set ${\cal H}$ of homoclinic orbits to ${\rm supp}(\mu)$ generates over $\re$ the relative homology
$H_{1}(M,U_{\varepsilon},\re)$.
In particular, there exist at least ${\rm \dim}\,H_{1}(M,U_{\varepsilon},\re)$
 homoclinic orbits to ${\rm supp}(\mu)$. 
  }

\vspace{.5cm}

S. Bolotin has shown in \cite[Theorem 3.4]{Bo1} and
\cite[Theorem 4.3]{Bo2} (cf. also \cite{Bo3})
 that if there exists a $C^{1+Lip}$ function
$f:M\to \re$ such that
\[L+c-df\geq 0,\]
then the set ${\cal H}$ of homoclinic orbits to ${\rm supp}(\mu)$ generates over $\re^+$ the relative homology
$H_{1}(M,U_{\varepsilon},\re)$.
In particular, he gets at least $2\,{\rm \dim}\,H_{1}(M,U_{\varepsilon},\re)$
 homoclinic orbits to ${\rm supp}(\mu)$, twice as much as we do in Corollary 1.
However we do not know if his condition
is generic.

Bolotin uses methods different from ours. To prove Theorem B
we consider finite coverings $M_{0}$ of $M$ whose group of deck transformations
is given by the quotient
of $H_{1}(M,U_{\varepsilon},\ze)/(\mbox{\rm torsion})$
 by a finite index 
subgroup.
Using that the lifted Lagrangian $L_{0}$ has the same critical value as $L$,
we conclude that the number of static classes of
$L_0$ must be finite.
Hence we can apply Theorem A to
 $L_{0}$ to deduce that the group generated by the homoclinic orbits to the set
of static orbits of $L$ coincides with $H_{1}(M,U_{\varepsilon},\re)$.

We note that the homoclinic orbits that we obtain in Theorem
B and Corollary 1 have energy $c$ but they are not semistatic
orbits of $L$ (cf. Theorem C). However, they are semistatic for
lifts of $L$ to suitable finite covers.


Using Corollary 1, we shall show in Section 6:

\vspace{.5cm}

\noindent {\bf Corollary 2.} {\it Let $M$ be a closed manifold with first Betti
number $\geq 2$.
 Given a
Lagrangian $L$ there exists a generic set ${\cal O}\subset C^{\infty}(M,\re)$
such that if $\psi\in{\cal O}$ the Lagrangian $L+\psi$ has a unique
minimizing measure in ${\cal M}^{0}(L+\psi)$ and this measure
is uniquely ergodic. When this measure is supported on a periodic orbit, this
orbit is hyperbolic and the stable and unstable manifolds have transverse homoclinic intersections. 
  }

\vspace{.5cm}

We say that a Lagrangian $L$ is {\it symmetric} if for all $(x,v)\in TM$,
$L(x,v)=L(x,-v)$. Note that if $L$ is symmetric and $\psi\in C^{\infty}(M,\re)$
then $L+\psi$ is also symmetric.

\vspace{.5cm}

\noindent {\bf Corollary 3.} {\it Let $M$ be a closed manifold with
first Betti number different from zero.
Given a symmetric Lagrangian $L$ there exists a generic
 set ${\cal O}\subset C^{\infty}(M,\re)$
such that if $\psi\in{\cal O}$, then $L+\psi$ has a unique
minimizing measure in ${\cal M}^{0}(L+\psi)$ and this measure
is supported on a hyperbolic fixed point
whose stable and unstable manifolds have transverse
homoclinic intersections. 
  }

\vspace{.5cm}

In \cite{Fa3} Albert Fathi has obtained independently
results which have a considerable overlap with Theorem B.
He defines a set ${\cal C}_{0}$ by
\[{\cal C}_{0}=\bigcup_{p}\,dp(\Sigma_{0}),\]
where the union is taken over all finite and
abelian Galois covers $p:M_{0}\to M$ and $\Sigma_{0}$ is the set
of semistatic orbits of the lift of $L$ to $M_{0}$.
He shows that the connected invariant set ${\cal C}_{0}$
is contained in $W^{s}(\hS)\cap W^{u}(\hS)$ and that for any
connected open set $V$ containing ${\cal C}_{0}$ one has
$H_{1}(TM,V,\ze)=0$. As a corollary, he also obtains
the existence of at least ${\rm \dim}\,H_{1}(M,U_{\varepsilon},\re)$
homoclinic orbits to the set of static orbits $\hS$ and without
assuming that $\hS$ contains only one static class.

At this point it seems useful to note
that there are various terminologies in the literature
for several of the concepts that we use here.
Fathi refers in \cite{Fa1,Fa2,Fa3,Fa4} to the closure of the union
of the support of minimizing measures as the {\it Aubry-Mather set}. 
What we call here the static and semistatic sets, Fathi calls
the {\it Peierls set} and the {\it Ma\~n\'e set} respectively.
As we mentioned before, semistatic and static curves are closely
 related to Mather's notions
of {\it c-minimal trajectories} and {\it regular c-minimal trajectories} respectively.
The terminology we follow in this paper is that of Ma\~n\'e in \cite{Ma}.


\vspace{.5cm}

{\it Acknowledgments:} The authors thank the IMPA for hospitality while this work was in progress.
G.P. Paternain thanks the ICTP for hospitality while this work was in progress.
Finally, we wish to thank the referee for pointing out a mistake in a previous
version of this manuscript.


\section{Critical values, static and semistatic curves}

Let $M$ be a closed connected manifold and
$L:TM\to\re$ a convex superlinear Lagrangian.

The action of the Lagrangian $L$ on
 an absolutely continuous curve $\ga:[a,b]\rightarrow M$ is defined by
\[A_{L}(\ga)=\int_{a}^{b}L(\ga(t),\dot{\ga}(t))\,dt.\]
Given two points, $x$ and $y$ in $M$ and $T>0$
 denote by ${\cal C}_{T}(x,y)$ the set of absolutely continuous curves $\ga:[0,T]\rightarrow M$, with $\ga(0)=x$ and $\ga(T)=y$.
For each $k\in \re$ we define the {\it action potential} $\Phi_{k}:M\times M\rightarrow \re$ by
\[\Phi_{k}(x,y)=\inf\{A_{L+k}(\ga):\;\ga\in\cup_{T>0}\, {\cal C}_T(x,y)\}.\]

\begin{Theorem}[Basic properties of the critical value \cite{Ma,CDI}]
There exists $c(L)\in \re$ such that
\begin{enumerate}
\item if $k<c(L)$, then $\Phi_{k}(x_{1},x_{2})=-\infty$, for all $x_{1}$ and $x_{2}$ in $M$;
\item if $k\geq c(L)$, then $\Phi_{k}(x_{1},x_{2})>-\infty$ for all $x_{1}$ and $x_{2}$ in $M$ and $\Phi_{k}$ is a Lipschitz function;
\item if $k\geq c(L)$, then
\[\p(x_{1},x_{3})\leq \p(x_{1},x_{2})+\p(x_{2},x_{3}),\]
for all $x_{1}$, $x_{2}$ and $x_{3}$ in $M$ and
\[\p(x_{1},x_{2})+\p(x_{2},x_{1})\geq 0,\]
for all $x_{1}$ and $x_{2}$ in $M$;
\item if $k>c(L)$, then for $x_{1}\neq x_{2}$ we have
\[\p(x_{1},x_{2})+\p(x_{2},x_{1})> 0.\]
\end{enumerate}               \label{basic}
\end{Theorem}

Observe that in general the action potential $\p$ is {\bf not} symmetric, however defining $d_{k}:M\times M\rightarrow \re$  by
\[d_{k}(x,y)=\p(x,y)+\p(y,x),\]
Theorem \ref{basic} says that $d_{k}$ is a metric for $k>c(L)$ and a pseudometric for $k=c(L)$.
The number $c(L)$ is called the {\it critical value} of $L$.

It is important for our purposes to indicate that the theorem above also hold for
coverings of $M$, i.e. suppose $\widehat{M}$ is a covering of $M$ with
 covering projection $p$. 
Take the lift of the Lagrangian $L$ to $\widehat{M}$ which is given by 
\[\widehat{L}(x,v)=L(p(x),dp(v)).\]
Then we define for each $k\in \re$ the action potential just as above
and the results hold for $\widehat{L}$. Thus we have a critical value for $\widehat{L}$.


Using the theorem it is straightforward to check that if
$M_{1}$ and $M_{2}$ are coverings of $M$ such that $M_{1}$
covers $M_{2}$, then
\begin{equation}
c(L_{1})\leq c(L_{2}),  \label{cubre}
\end{equation}
where $L_{1}$ and $L_{2}$ denote the lifts of the Lagrangian $L$ to $M_{1}$
and $M_{2}$ respectively.
Also we have the following lemma.

\begin{Lemma}
If $M_{1}$ is a {\it finite} covering of $M_{2}$ then
$c(L_{1})= c(L_{2}). $  \label{igual}
\end{Lemma}

\PROOF We know that $c(L_{1})\leq c(L_{2})$. Suppose that
the strict inequality holds and let $k$ be such that
$c(L_{1})<k<c(L_{2})$. Hence there exists a closed curve
$\ga$ in $M_{2}$ with negative $(L_{2}+k)$-action. Since $M_{1}$
is a finite covering of $M_{2}$ some iterate of $\ga$ lifts to
a closed curve in $M_{1}$ with negative $(L_{1}+k)$-action
which contradicts $c(L_{1})<k$.

\qed

Note that for every absolutely continuous curve $\ga:[a,b]\to M$
and all $k\geq c(L)$ Theorem \ref{basic} implies that:
\begin{equation}
A_{L+k}(\ga)\geq \Phi_{k}(\ga(a),\ga(b))\geq -\Phi_{k}(\ga(b),\ga(a)).\label{se'}
\end{equation}
Set $c=c(L)$. We say that an absolutely continuous curve
 $\ga:[a,b]\to M$ is {\it semistatic} if
\[A_{L+c}(\ga|_{[t_{0},t_{1}]})=\Phi_{c}(\ga(t_{0}),\ga(t_{1})),\]
for all $a<t_{0}\leq t_{1}<b$; and that is {\it static}
if 
\[A_{L+c}(\ga|_{[t_{0},t_{1}]})=-\Phi_{c}(\ga(t_{1}),\ga(t_{0})),\]
for all $a<t_{0}\leq t_{1}<b$.
Clearly, by (\ref{se'}) a static curve is semistatic.
One could also say that $\ga|_{[a,b]}$ is static
if it is semistatic and for all $a<t_{0}\leq t_{1}<b$,
 $d_{c}(\ga(t_0),\ga(t_1))=0$.
Semistatic curves are solutions of the Euler-Lagrange equation because
of their minimizing properties. Also it is not hard
to check that semistatic curves have energy precisely $c$ \cite{Ma, CDI}.

Given a vector $v\in TM$ we shall denote by $x_v:\re\to M$ the
solution of the Euler-Lagrange equation with $\dot{x}(0)=v$.

The set of vectors $v$ in $TM$ that give rise to static curves $x_v:\re\to M$ is an invariant set that we shall denote by $\hS:=\hS(L)$.
Similarly, 
the set of vectors $v$ in $TM$ that give rise to semistatic curves $x_v:\re\to M$ is an invariant set that we shall denote by $\Sigma:=\Sigma(L)$.
As we mentioned before $\hS\subset \Sigma$.
The continuity properties of $A_{L+c}$ and $\Phi_{c}$ imply that
$\Sigma$ and $\hS$ are closed sets.

\begin{Lemma}\label{lifts}
Let $p:M_{1}\to M_{2}$ be a covering such that $c(L_{1})=c(L_{2})$.
Then any lift of a semistatic curve of $L_{2}$ is a semistatic curve
of $L_{1}$. Also the projection of a static curve of $L_{1}$ is
a static curve of $L_{2}$. If in addition, $p$ is a finite covering,
then any lift of a static curve of $L_{2}$ is a static curve
of $L_{1}$.
\end{Lemma}

\PROOF Observe first that for any $k\in\re$ we have that
\[\Phi_{k}^{1}(x,y)\geq \Phi_{k}^{2}(px,py),\]
for all $x$ and $y$ in $M_{1}$. Hence if we write $c=c(L_{1})=c(L_{2})$
we have
\begin{equation}
\Phi_{c}^{1}(x,y)\geq \Phi_{c}^{2}(px,py),\label{eins}
\end{equation}
for all $x$ and $y$ in $M_{1}$.

Suppose now that $x_{2}:\re\to M_{2}$ is a semistatic curve of $L_{2}$
and let $x_{1}:\re\to M_{1}$ be any lift of $x_{2}$ to $M_{1}$.
Using (\ref{eins}) and the fact that $x_{2}$ is semistatic we
have for $s\leq t$,

\begin{align*}
\Phi_{c}^{1}(x_{1}(s),x_{1}(t)) 
    & \leq A_{L_{1}+c}(x_{1}|_{[s,t]})=A_{L_{2}+c}(x_{2}|_{[s,t]})\\
    & =\Phi_{c}^{2}(x_{2}(s),x_{2}(t))\leq \Phi_{c}^{1}(x_{1}(s),x_{1}(t)).
\end{align*}
Hence $x_{1}$ is semistatic for $L_{1}$.

Suppose now that $x_{1}:\re\to M_{1}$ is a static curve of $L_{1}$
and let $x_{2}:\re\to M_{2}$ be $p\circ x_{1}$.
Using (\ref{eins}) and the fact that $x_{1}$ is static we
have for $s\leq t$,
\begin{align*}
-\Phi_{c}^{1}(x_{1}(t),x_{1}(s))&=\Phi_{c}^{1}(x_{1}(s),x_{1}(t)) 
  = A_{L_{1}+c}(x_{1}|_{[s,t]})=A_{L_{2}+c}(x_{2}|_{[s,t]})\\
    & \geq \Phi_{c}^{2}(x_{2}(s),x_{2}(t)) \geq -\Phi_{c}^{2}(x_{2}(t),x_{2}(s)) \geq -\Phi_{c}^{1}(x_{1}(t),x_{1}(s)).
\end{align*}
Hence $x_{2}$ is static for $L_{2}$.

Suppose now that $p$ is a {\it finite} covering and let
$x_{2}:\re\to M_{2}$ be a static curve of $L_{2}$. Let
$x_{1}:\re\to M_{1}$ be any lift of $x_{2}$ to $M_{1}$. Since $x_{2}$
is static, given $s\leq t$ and $\e>0$, there exists a curve
$\alpha:[0,T]\to M_{2}$ with $\alpha(0)=x_{2}(t)$, $\alpha(T)=x_{2}(s)$
such that 
\[A_{L_{2}+c}(x_{2}|_{[s,t]})+A_{L_{2}+c}(\alpha)\leq \e.\]
Since $p$ is a finite covering, there exists a positive integer $n$,
bounded from above by the number of sheets of the covering,
such that the $n$-th iterate
of $x_{2}|_{[s,t]}*\alpha$ lifts to $M_{1}$ as a closed curve.
Hence, there exists a curve $\beta$ joining $x_{1}(t)$ to $x_{1}(s)$
such that
\[A_{L_{1}+c}(x_{1}|_{[s,t]})+A_{L_{1}+c}(\beta)\leq n\e,\]
and thus $x_{1}$ is static for $L_{1}$.

\qed

 \section{Proof of Theorem A}
  We shall endow $M$ with a Riemannian metric and we consider in $TM$
  the associated Sasaki metric. Let $d_{M}$ and $d_{TM}$ be the corresponding    distance functions of these Riemannian metrics.
  Given $v\in TM$ denote by $\a(v)$ and $\om(v)$ its $\a$ and
  $\om$-limits respectively.
  We recall the following:
  
  \begin{Lemma}[\cite{CDI}]
   If $v\in \Sigma$ is semistatic, then $\a(v)\subset\widehat{\Sigma}$
  and $\om(v)\subset \widehat{\Sigma}$. Moreover, $\a(v)$ and $\om(v)$
  are each included in a static class.\label{wlim}
  \end{Lemma}
 
 
 Set
 $$\Sigma^\e :=\{w\in TM \,\vert\, x_w:[0,\e)\to M \text{ or }
             x_w:(-\e,0]\to M \text{ is semistatic }\}.$$
 
 
 \begin{Theorem}[Graph Property, see \cite{Ma,CDI} and \cite{M'}]
  For all $p\in \pi(\hS)$ there exists a unique $\xi(p)\in T_pM$
  such that $(p,\xi(p))\in\Sigma^\e$, in particular
  $(p,\xi(p))\in\widehat{\Sigma}$ and $\widehat{\Sigma}=
  \text{graph}(\xi)$. Moreover, there exist positive constants
 $\eta$ and $K$ such that if $(p,v)\in \hS$, $(q,w)\in \Sigma^{\e}$
and $d_{M}(p,q)<\eta$ then 
\[d_{TM}((p,v),(q,w))<K\,d_{M}(p,q).\] 
In particular, the map $\xi:\pi(\hS)\to \Sigma$  
  is Lipschitz.  \label{GraphProperty}
 \end{Theorem}
  
 
 Using the Graph Property we can define an equivalence relation on $\widehat\Sigma$ by
  $$
  u,v\in \widehat{\Sigma}, \quad u\equiv v \iff d_c(\pi(u),\pi(v)) =0.
  $$
  The equivalence classes are called {\it static classes}.
  Let $\boldsymbol{\La}$ be the set of static classes. Define a
  reflexive partial order $\preceq$ in $\boldsymbol{\La}$ by
  \begin{itemize}
  \item[(a)] $\preceq$ is reflexive.
  \item[(b)] $\preceq$ is transitive.
  \item[(c)] If there is $v\in\Sigma$ with $\a(v)\subseteq\La_i$ and
             $\om(v)\subseteq\La_j$, then $\La_i\preceq\La_j$.
  \end{itemize}            
  
 
 Let us begin with the proof of the theorem.
 We shall prove in Proposition \ref{ConnectedClass} below that
 the static classes are connected. Hence if we assume
 that there are only finitely many of them, 
  the connected components of $\hS$ are
  finite and must coincide with the static classes.
  For $\e>0$, let $\hS(\e)$ be the $\e$-neighborhood of
  $\hS$, i.e.
  $$
  \hS(\e) :=\{ \, v\in TM \, \vert \,
  d_{TM}(v,\hS)<\e\,\}.
  $$
  Fix $\e>0$ small enough such that $\e<\eta$ where $\eta$ is the positive
  constant given by Theorem \ref{GraphProperty} and such that
  the connected components of
  $\hS(\e)$ are the $\e$-neighborhoods of the static classes.
  Thus for $0<\de<\e$, $\hS(\de)=\sum_{i=1}^{N(\e)}\La_i(\de)$,
  where $\La_i(\de)$ are disjoint open sets containing exactly one
  static class and the number of components $N(\e)$ is fixed for all
  $0<\de<\e$.
  
  Now suppose that the theorem is false. This means that there exists
  $\La\in\boldsymbol{\La}$ such that the following two sets are not
  empty:
  $$
  {\Bbb A}:=\bigcup_{\{\,\La_j\in\boldsymbol{\La}\,\vert\,
  \La\preceq\La_j\,\}}\La_j
  \quad , \quad
  {\Bbb B} :=\bigcup_{ \{\,\La_j\in\boldsymbol{\La}\,\vert\,
  \La\not\preceq\La_j\,\}}\La_j.
  $$
  
  Given $v\in\Sigma$ with $\a(v)\subseteq{\Bbb A}$ and $0<\de<\e$,
  define inductively $s_k(v)$, $t_k(v)$, $T_k(v)$ as follows. Let
  \begin{align*}
  s_1(v) &:= \inf \{\, s\in\re\,\vert\, f_s(v)\notin{\Bbb A}(\e)\,\}
  \in\re\cup\{+\infty\}.\\
  \intertext{If $s_k(v)<+\infty$, $k\geq 1$, define }
  t_k(v) &:= \sup\{\, t<s_k(v)\,\vert\, f_t(v)\in{\Bbb A}(\de)\,\},  \\ 
  T_k(v) &:= \inf\{\,t>s_k(v)\,\vert\,f_t(v)\in{\Bbb A}(\de)\,\}.   \\          
  \intertext{Observe that $s_k(v)<+\infty$ implies that
  $T_k(v)<+\infty$ because by the definition of ${\Bbb B}$ and the 
  transitivity of $\preceq$ we have that $\om(v)\subseteq {\Bbb A}$.
  Define}
  A_k=A_k(\de) &:= \sup \{\,\vert\,T_k(v)-t_k(v)\,\vert\;:\;
                 v\in\Sigma,\;\a(v)\subseteq{\Bbb A},\,
		 s_k(v)<+\infty\,\},
  \end{align*}
  if $s_k(v)=+\infty$ for all  $v\in\Sigma$ with $\a(v)\subseteq{\Bbb
  A}$, write $A_\ell(\de)\equiv 0$ for all $\ell\ge k$. Now set:
  \[s_{k+1}(v):= \inf\{\,s>T_k(v)\,\vert\,f_t(v)\notin{\Bbb A}(\e)\,\}.\]
  Observe that $s_k(v)$, $t_k(v)$ and $T_k(v)$ are invariant under $f_{t}$.
  
  \bigskip
  
  We split the rest of the proof of Theorem A into
  the following claims:
 
  \medskip
  
  \noindent{\bf Claim 1.} 
  {\it $A_k(\de)<+\infty$ for all $k=1,2,\ldots$ and all $0<\de<\e$.}
  
  \medskip
  
  Define
 % $$
  %{\Bbb M} := \{\, f_t(v)\,\vert\,
 % t\in\re,\;v\in\Sigma,\;\a(v)\subseteq{\Bbb A}\,\}.
 % $$
 $${\Bbb M} := \{\,v\,\vert\,\;v\in\Sigma,\;\a(v)\subseteq{\Bbb A}\,\}.
  $$
  
 \medskip
 
  \noindent{\bf Claim 2.}
  {\it\begin{itemize}
  \item[(a)] $\ov{\Bbb M}\cap{\Bbb B}\ne \nulo$.
  \item[(b)] $\limsup_{k}A_k(\de)=\sup_k A_k(\de)=+\infty$.
  \end{itemize}}
  
  \medskip
  
  \noindent{\bf Claim 3.}
  {\it There exist sequences $v_n\in\Sigma$, $0<s_n<t_n$ such that
  $v_n\to u_1\in{\Bbb A}$, $f_{s_n}(v_n)\to u_2\notin{\Bbb A}(\e)$,
  $f_{t_n}(v_n)\to u_3\in{\Bbb A}$ and $d_c(\pi u_1, \pi u_3) = 0$.} 
  
  \medskip
  
  We now use Claim~3 to complete the proof of Theorem A. If
  $u_1\in\La_j\subseteq{\Bbb A}$, we shall prove that
  $u_2\in\La_j\setminus{\Bbb A}(\e)$, obtaining a contradiction
  and thus proving Theorem A. It is enough to show
  that $d_c(\pi u_1,\pi u_2)=0$. Indeed
  \begin{align*}
  d_c(\pi u_1,\pi u_2) 
     &= \Phi_c(\pi u_1,\pi u_2) + \Phi_c(\pi u_2,\pi u_1) \\
     &\le \Phi_c(\pi u_1,\pi u_2) + \Phi_c(\pi u_2, \pi u_3)
          + \Phi_c(\pi u_3, \pi u_1) \\
     &= \lim_n\bigl[\Phi_c(\pi v_n,\pi f_{s_n}(v_n))
          + \Phi_c(\pi f_{s_n}(v_n),\pi f_{t_n}(v_n))\bigr]
          + \Phi_c(\pi u_3,\pi u_1) \\
     &=\lim_n \Phi_c(\pi v_n, \pi f_{t_n}(v_n))
          + \Phi_c(\pi u_3,\pi u_1) \\
     &= d_c(\pi u_1, \pi u_3) = 0,
   \end{align*}
   where the fourth equality holds because $v_n$ is a semistatic
   vector.
   
   \qed           
   
   
   \medskip
  
   We need the following
 
 \begin{Lemma}\cite[Corollary 1.4]{CDI}\label{bounded}
   
   There exists $A>0$ such that if $p,q\in M$ and $x\in {\cal C}_{T}(p,q)$
   satisfy
   \begin{itemize}
   \item[(a)] $A_L(x)=\min\{\,A_L(y)\,\vert\, y\in {\cal C}_{T}(p,q)\,\}$;
   \item[(b)] $A_{L+c}(x) < \Phi_c(p,q) + d_M(p,q)$,
   \end{itemize}
   then  $| \dot{x}(t)| < A$ for all $t\in[0,T]$.
   \end{Lemma} 
   
   \bigskip
   
   \noindent{\em Proof of Claim 1.}
    Suppose that $A_i<+\infty$ for $i=1,\ldots,k-1$ and $A_k=+\infty$.
   The case $k=1$ is similar. Then there exists $v_n\in\Sigma$, with
   $\a(v_n)\subset{\Bbb A}$ and $T_k(v_n)-t_k(v_n)\to +\infty$. We can
   assume that $t_k(v_n)=0$ and that $v_n$ converges ($\Sigma$ is compact). Let
   $u= \lim_n v_n\in\partial{\Bbb A}(\de)$. Then for all $n$ we have
   \begin{equation}
   m\{\, t<0\,\vert\,f_t(v_n)\notin {\Bbb
   A}(\e)\,\}\le\sum_{i=1}^{k-1}A_i\,,\label{medida}
   \end{equation}
   where $m$ is the Lebesgue measure on $\re$. We claim that
   $\a(u)\subset{\Bbb A}$. To prove the claim it suffices to show
   that there is a sequence $r_{m}\to -\infty$ such that $f_{r_{m}}(u)\in
   \overline{{\Bbb A}(\e)}$. (Recall that $\alpha(u)$ must be contained in a
   unique static class by Lemma \ref{wlim}.) Suppose that such a sequence does    not exist. This means that there exists $R\leq 0$ such that for all
   $t\leq R$, $f_{t}(u)\not\in\overline{{\Bbb A}(\e)}$. Since $v_{n}\to u$, there   exists $n$ sufficiently large for which
   $f_{t}(v_{n})\not\in\overline{{\Bbb A}(\e)}$ for all $t\in[R-\sum_{i=1}^{k-1}A   _i-2,R-1]$. This contradicts (\ref{medida}).

   Since $f_t(v_n)\notin{\Bbb A}(\e)$
   for $0<t<T_k(v_n)$ and $T_k(v_n)\to+\infty$, then
   $f_t(u)\notin{\Bbb A}(\e)$ for all $t>0$ and hence
   $\om(u)\subseteq{\Bbb B}$. But then the orbit of $u$ contradicts the
   definition of ${\Bbb B}$.
   
   \qed
   
    
   \bigskip
   
   {\noindent \em Proof of Claim 2.}
   
   \smallskip
   
   (a) Let $p\in\pi{\Bbb A}$, $q\in\pi{\Bbb B}$. For $n>0$, let
   $x_{v_n}:[a_n,b_n]\to M$ be a solution of (E-L) such that
   $x_{v_n}(a_n)=p$, $x_{v_n}(b_n)=q$ and
   $$
   A_{L+c}(x_{v_n})\le \Phi_c(p,q) + \tfrac 1n.
   $$
   This implies that
   \begin{equation}\label{E:m.1}
   A_{L+c}\left(x_{v_n}\vert_{[s,t]}\right)\le
   \Phi_c\bigl(x_{v_n}(s),x_{v_n}(t)\bigr)+\tfrac 1n
   \end{equation}
   for all $a_n\le s\le t\le b_n$. We can assume that
   $$
   \inf\{\, s>a_n\,\vert\,x_{v_n}(s)\in\pi\,{\Bbb B}(\de)\,\}=0,
   $$
   and that the sequence $v_n$ converges (cf. Lemma~\ref{bounded}). 
   Let $u=\lim_n v_n\in\pi^{-1}(\partial\,\pi\,{\Bbb B}(\de))$. 
   Taking limits in
   \eqref{E:m.1} we obtain that $x_u\vert_{[s,t]}$ is
   semistatic for all $\lim\inf_n a_n \le s\le t \le \lim\sup_n b_n$.
   
   Any limit point $w$ of $\dot{x}_{v_n}(a_n)=f_{a_n}(v_n)$ satisfies
   $\pi(w)=p\in\pi{\Bbb A}$, and by the Graph
   Property (Theorem~\ref{GraphProperty}), $w\in{\Bbb A}$. Similarly, any limit
   point of $f_{b_n}(v_n)$ is in ${\Bbb B}$. Since ${\Bbb A}\cup {\Bbb
   B}$ is invariant and $u\notin{\Bbb A}\cup{\Bbb B}$, then $\lim_n
   a_n=-\infty$, $\lim_n b_n=+\infty$. Hence $u\in\Sigma$. Since
   $f_t(v_n)\notin{\Bbb B}(\de)$ for all $a_n\le t<0$ and $a_n\to
   -\infty$, then $f_t(u)\notin{\Bbb B}(\de)$ for all $t<0$. Hence
   $\a(u)\subseteq{\Bbb A}$ and thus $u\in {\Bbb M}$. Since 
   $u\in\pi^{-1}(\partial\,\pi\,{\Bbb B}(\de))$ there exists $z\in{\Bbb B}$
   such that $d_{M}(\pi(u),\pi(z))\leq \de$. Since $z\in \hS$ and $u\in\Sigma$    by Theorem \ref{GraphProperty} we have
   \[d_{TM}((\pi(u),u),(\pi(z),z))\leq K\,\de.\]
   Thus $u\in{\Bbb M}\cap {\Bbb B}(K\,\de)$. Letting $\de\to 0$, we obtain that
   $\ov{\Bbb M}\cap{\Bbb B}\ne\nulo$.
   
                  
   \medskip               
                  
   (b) By Claim~1 it is enough to show that $\sup_k A_k(\de)=+\infty$.
   If $\sup_k A_k(\de)<T$, then ${\Bbb M}\subseteq{\Bbb M}(\de,T)$,
   where ${\Bbb M}(\de,T)$ is the compact set given by:
   $$
   {\Bbb M}(\de,T)=\{\, v\in\Sigma\,\vert\, f_{[-T,T]}(v)\cap
   \ov{{\Bbb A}(\de)}\ne \nulo\,\}=f_{[-T,T]}\left(\overline{{\Bbb A}(\de)}\cap   \Sigma\right).
   $$
   Note that ${\Bbb M}(\de,T)\cap{\Bbb
   B}=\nulo$, because ${\Bbb B}$ is invariant and ${\Bbb
   B}\cap{\Bbb A}(\de)=\nulo$. On the other hand, since ${\Bbb M}\subseteq
   {\Bbb M}(\de,T)$ we have
   $\overline{{\Bbb M}}\cap {\Bbb B}\subseteq{\Bbb M}(\de,T)
   \cap{\Bbb B}=\nulo$. 
   This contradicts item (a).
   
   \qed
   
   \medskip
   
   \noindent{\em Proof of Claim~3.}
   Given $0<\de<\e$, by Claim~2(b) there exists $k>N(\e)$ such that
   $A_k(\de)>0$. Hence there is $v=v_\de\in\Sigma$ with
   $\a(v)\subset{\Bbb A}$, such that the orbit of $v$ leaves ${\Bbb
   A}(\e)$ and returns to ${\Bbb A}(\de)$ at least $k$ times. Since
   $k>N(\e)$ there is one component $\La_j(\de)\subseteq{\Bbb A}(\de)$
   with two of these returns, i.e. there exist
   $\tau_1(\de)<s(\de)<\tau_2(\de)$ with $f_{\tau_1}(v)\in\La_j(\de)$,
   $f_s(v)\notin{\Bbb A}(\e)$ and $f_{\tau_2}(v)\in\La_j(\de)$. 
   We can choose $v_\de$ so that $\tau_1(\de)=0$. Now,
   there exists a sequence $\de_n\downarrow 0$
   such that the repeated component
   $\La_j\subset\La_j(\de_n)$ is always the same. Let
   $s_n:=s(\de_n)$, $t_n:=\tau_2(\de_n)$, $v_{n}:=v_{\de_{n}}$
   and choose a subsequence 
   such that $v_{n}$, $f_{s_n}(v_n)$ and
   $f_{t_n}(v_n)$ converge. Let $u_1=\lim_n v_n\in \cap_n
   \La_j(\de_n)=\La_j$, $u_3=\lim_n f_{t_n}(v_n)\in\La_j$ and
   $u_2=\lim_n f_{s_n}(v_n)\notin{\Bbb A}(\e)$. Since
   $u_1,u_3\in\La_j$, then $d_c(\pi u_1,\pi u_3)=0$.
   
   \qed
   
   
   \bigskip
 
  \begin{Proposition}\label{ConnectedClass}
  Every static class is connected.
  \end{Proposition}
   
   \PROOF   
  Let $\La$ be a static class and suppose that it is not connected. 
    Let $U_1,U_2$ be disjoint open sets such that $\La\subseteq U_1\cup U_2$
    and $\La\cap U_i\ne\nulo$, $i=1,2$. Let $p_i\in\pi(U_i\cap\La)$,
    $i=1,2$.
  Since $U_{1}$ and $U_{2}$ are disjoint sets we can take
  a solution
  $x_{v_n}:[a_n,b_n]\to M$, $a_n<0<b_n$ of (E-L)
  such that $x_{v_n}(0)\notin \pi(U_1\cup U_2)$,
    $x_{v_n}(a_n)=p_1$, $x_{v_n}(b_n)=p_2$ and
 \begin{equation}
    A_{L+c}(x_{v_n})\le\Phi_c(p_1,p_2)+\tfrac 1n.  \label{zwei}
 \end{equation}
    Let $u$ be a limit point of $v_n$,
    then $x_u:\re\to M$ is semistatic (see the proof of Claim~2 item (a)). 
 Then, for $a_n\le s\le t\le
    b_n$,
 $$d_c(p_1,p_2) 
      \leq\Phi_c(p_1,x_{v_n}(s))+\Phi_c(x_{v_n}(s),x_{v_n}(t))
           +\Phi_c(x_{v_n}(t),p_2)+\Phi_c(p_2,p_1),$$
 therefore
    \begin{align*}
 d_c(p_1,p_2) 
      &\leq\liminf_n\left[\Phi_c(p_1,x_{v_n}(s))
             +\Phi_c(x_{v_n}(s),x_{v_n}(t))+\Phi_c(x_{v_n}(t),p_2)\right]
             +\Phi_c(p_2,p_1)  \\
      &\leq \liminf_{n} A_{L+c}(x_{v_{n}})+\Phi_{c}(p_{2},p_{1})\\
      &\leq d_{c}(p_{1},p_{2})=0,
 \end{align*}
 where in the last inequality we used (\ref{zwei}).
 Hence
 $$\Phi_c(p_1,x_{u}(s))+\Phi_c(x_{u}(s),x_{u}(t))
           +\Phi_c(x_{u}(t),p_2)+\Phi_c(p_2,p_1)=0.$$
    Combining the last equation with the triangle inequality
 we obtain
    $$
    d_c(x_u(s),x_u(t))\le \Phi_c(x_u(s),x_u(t))
          +\left[\Phi_c(x_u(t),p_2)+\Phi_c(p_2,p_1)
          +\Phi_c(p_1,x_u(s))\right]
          =0.
    $$
  So that $u\in\widehat{\Sigma}$. Moreover, for $s=0,t=1$:
  $$
  d_c(x_u(0),p_1)\le\Phi_c(p_1,x_u(0))
        +\left[\Phi_c(x_u(0),x_u(1))+\Phi_c(x_u(1),p_2)
        +\Phi_c(p_2,p_1)\right]=0.
  $$
  Hence $x_{u}(0)\in\pi(\La)$. On the other hand $x_{u}(0)\notin \pi(U_{1}\cup U_{2})$.
  This contradicts the fact that $\La\subseteq U_1\cup U_2$.
  
  \qed      
                
          




\section{Proof of Theorem B}



Let $U\df U_{\varepsilon}$ denote the $\varepsilon$-neighborhood of $\pi(\hS(L))$,
where $\hS(L)$ is the set of static vectors of $L$.
Since we are assuming that $\hS(L)$ contains only one static class, the set $U$ is also connected
for small $\varepsilon$. Let $i:U\to M$ be the inclusion map. 
The vector space $H_{1}(M,U,\re)$ is isomorphic to the quotient
of $H_{1}(M,\re)$ by $i_{*}(H_{1}(U,\re))$.

Let $G$ be the quotient of $H_{1}(M,U,\ze)$ by its torsion part. Since $G$ 
is free we can write $G=\ze\oplus\overset{k}{\ldots}\oplus\ze$
where $k={\rm \dim}\,H_{1}(M,U,\re)$. The group $G$ can be seen as a lattice
in $H_{1}(M,U,\re)$.
Let $J$ be a finite index subgroup of $G$. 
There is a surjective homomorphism $j:G\to G/J$ given by the projection.

If we take the Hurewicz map
\[\pi_{1}(M)\mapsto H_{1}(M,\ze),\]
and we compose it with the surjective homomorphisms $H_{1}(M,\ze)\mapsto H_{1}(M,U,\ze)$, $H_{1}(M,U,\ze)\mapsto G$ and
$j:G\to G/J$, we obtain a surjective homomorphism
\[\pi_{1}(M)\mapsto G/J,\]
whose kernel will be the fundamental group of a finite Galois covering
$M_{0}$ of $M$ with covering projection map $p:M_{0}\to M$
and group of deck transformations given by the finite abelian group $G/J$.

Observe that $G/J$ acts transitively
and freely on the set of connected components of $p^{-1}(U)$ which
coincides with the set of connected components of $p^{-1}(\pi(\hS(L)))$.
Therefore we have

\begin{Lemma}There is a one to one correspondence between elements
in $G/J$ and connected components of $p^{-1}(\pi(\hS(L)))$.  \label{components}
\end{Lemma}

Observe that to each homoclinic orbit $x:\re\to M$
 to $\hS(L)$ we can associate
a homology class in $G$. Indeed, since there exists $t_{0}>0$ such that
for all $t$ with $|t|\geq t_{0}$, $x(t)\in U$, the class of
$x|_{[-t_{0},t_{0}]}$ defines an element in $H_{1}(M,U,\ze)$ and hence in $G$.
Let us denote by ${\cal H}$ the subset of $G$ given by all the classes
corresponding to homoclinic orbits to $\hS(L)$.


\begin{figure}

\special{psfile=bump1.eps hoffset=70 hscale=50 vscale=54 angle=-90}

\vspace{7cm} 

\caption{Creating homoclinic connections with finite coverings and
Theorem A.}

\label{fig2}

\end{figure}

\begin{Lemma} For any $J$ as above, the image of $\langle{\cal H}\rangle$
under $j$ is precisely $G/J$.  \label{j}
\end{Lemma}

\PROOF Let $L_{0}$ denote the lift of the Lagrangian $L$ to $M_{0}$.
Observe first that by Lemma \ref{igual}, $c(L)=c(L_{0})$ and therefore
 by Lemma \ref{lifts} we have
\begin{equation} 
\pi_{0}(\hS(L_{0}))= p^{-1}(\pi(\hS(L))),  \label{Igualdad}
\end{equation}
where $\pi_{0}:TM_{0}\to M_{0}$ is the canonical projection
of the tangent bundle $TM_{0}$ to $M_{0}$.

Let us prove now that $L_{0}$ satisfies the hypothesis of Theorem A, that
is, the number of static
classes of $L_{0}$ is finite. In fact, we shall show that the projection
to $M_{0}$ of a static
class of $L_{0}$ coincides with a connected component of
$p^{-1}(\pi(\hS(L)))$.
Using (\ref{Igualdad}) and Proposition \ref{ConnectedClass} we 
see that the projection of a static class of $L_0$ to $M_0$
must be contained in a single connected component of $p^{-1}(\pi(\hS(L)))$.
Hence, it suffices to show that if $x$ and $y$ belong to a
connected component of $p^{-1}(\pi(\hS(L)))$
then $d^{0}_{c}(x,y)=0$. Since we are assuming that $\hS(L)$ contains
only one static class we have that $d_{c}(px,py)=0$.
Since $p:M_{0}\to M$ is a finite covering there are lifts $x_{1}$ of $px$
and $y_{1}$ of $py$ such that $d_{c}^{0}(x_{1},y_{1})=0$.
Since static classes are connected $x_{1}$ and $y_{1}$ must belong to
the same connected component of $p^{-1}(\pi(\hS(L)))$ and thus
there is a covering transformation taking $x_{1}$ into $x$ and $y_{1}$
into $y$ which implies that $d^{0}_{c}(x,y)=0$ as desired.

Now Theorem A and (\ref{Igualdad}) imply that every covering transformation
in $G/J$ can be written as the composition of covering transformations
that arise from elements in ${\cal H}$, that is, 
$j(\langle{\cal H}\rangle)=G/J$.


\qed



We shall need the following algebraic lemma.

\begin{Lemma}Let $G=\ze\oplus\overset{k}{\ldots}\oplus\ze$.
Given a finite index subgroup $J\subset G$ let us denote by
$j:G\to G/J$ the projection homomorphism.

Let $A$ be a subgroup of $G$. If $A$ has the property
that for all $J$ as above $j(A)=G/J$, then
$A=G$.  \label{fato}
\end{Lemma}

\PROOF The hypothesis readily implies that
\begin{equation}
A/A\cap J\;\mbox{ is isomorphic to}\;G/J.          \label{isom}
\end{equation}
\begin{itemize}
\item If the rank of $A$ is strictly less than the rank of $G$,
one can easily construct a subgroup $J\subset G$ with finite index
such that $A\subseteq J$ and $G/J\neq\{0\}$.
But this contradicts (\ref{isom}) because $A/A\cap J=\{0\}$.
\item If the rank of $A$ equals the rank of $G$, then $A$ has finite
index in $G$ and by (\ref{isom}) $G/A=\{0\}$ and thus $G=A$.
\end{itemize}

\qed 




Observe now that any set ${\cal H}$ of a free abelian group $G$ of rank $k$ such that the group generated by ${\cal H}$ is $G$ must have at least
$k$ elements.
Therefore if we combine Lemma \ref{j} and Lemma \ref{fato} with
$\langle{\cal H}\rangle=A$ we deduce
that the set ${\cal H}$ of classes corresponding to homoclinic orbits
generates $G$ and
must have at least $k$ elements thus concluding the proof
of Theorem B.
\qed

\section{Proof of Theorem C}

To prove the theorem we shall show first several
lemmas. We will use the following notation:

 \begin{itemize}
  \item $\M^{0}(L)=$ minimizing measures of $L$;
  \item $\S(L)=$ semistatic vectors of $L$;
  \item $\hS(L)=$ static vectors of $L$;
  \item $\La(L)=$ closure of $\bigcup_{\mu\in \M^{0}(L)}\supp(\mu)$.
  \end{itemize} 

Recall that we always have $\La(L)\subseteq \hS(L)\subseteq \S(L)$. 

\begin{Lemma}The function 
$C^\infty(M,\re)\ni\psi\mapsto c(L+\psi)$ is continuous.
\label{uno}
\end{Lemma}

\PROOF Suppose that $\psi_{n}\to\psi$ and let $c_{n}:=c(L+\psi_{n})$
and $c:=c(L+\psi)$. We will prove that $c_n\to c$.

Fix $\e>0$. Since $c-\e<c$, by the definition of critical value
there exists a closed curve $\ga:[0,T]\to M$ such that
\[A_{L+\psi+c-\e}(\ga)<0,\]
hence for all $n$ sufficiently large
\[A_{L+\psi_{n}+c-\e}(\ga)<0,\]
therefore for all $n$ sufficiently large
\[c-\e<c_n,\]
and thus
\[c-\e\leq \liminf_n c_n.\]
Since $\e$ was arbitrary we have
\[c\leq \liminf_n c_n.\]

Let us show now that $\limsup_n c_n\leq c$. Suppose that
$c<\limsup_n c_n$. Take $\e$ such that
\begin{equation}
c<c+\e<\limsup_n c_n.
\label{ch}
\end{equation}
Since $\psi_{n}\to\psi$, there exists $n_0$ such that
for all $n\geq n_0$,
\begin{equation}
-\e\leq\psi_{n}-\psi\leq\e.
\label{ch1}
\end{equation}
By (\ref{ch}), there exists $m\geq n_0$ such that
\[c<c+\e<c_m.\]
By the definition of critical value
there exists a closed curve $\ga:[0,T]\to M$ such that
\[A_{L+\psi_{m}+c+\e}(\ga)<0,\]
and hence using (\ref{ch1}) we have
\[A_{L+\psi+c}(\ga)\leq A_{L+\psi_{m}+c+\e}(\ga)<0,\]
which yields a contradiction to the definition of the
critical value $c$.


\qed

This proof also shows that $L\mapsto c(L)$ is continuous
if we endow the set of Lagrangians $L$ with the
topology of uniform convergence on compact subsets of $TM$.


\begin{Lemma}$\lim_{\psi\to 0}\S(L+\psi)\subset
\S(L)$, where $\lim_{\psi\to 0}\S(L+\psi)$ is the set of accumulation
points of sequences $v_n\in\S(L+\psi_n)\subset TM$ with
$\psi_n\to 0$.
\label{dos}
\end{Lemma}


\PROOF  Let $\psi_n\to 0$ and $v_n\in \S(L+\psi_n)$ with $v_n\to v$.
     Let $T>0$ and write $x_{v_n}(t):=\pi_\circ f^n_t(v_n)$,
     $x_v(t):=\pi_\circ f_t(v)$, $x_n=x_{v_n}(-T)$, $y_n=y_{v_n}(T)$,
     $x=x_v(-T)$, $y=x_v(T)$, $c_n=c(L+\psi_n)$ and $c=c(L)$,
     where $f^n_t$ and $f_t$  are the Euler-Lagrange flows
     of $L+\psi_n$ and $L$ respectively. Then
     \begin{equation}\label{cvs.1}
     \Phi_c(x,y)\le A_{L+c}\bigl(x_v\vert_{[-T,T]}\bigr)
     =\lim_n A_{L+\psi_n+c_n}\bigl(x_{v_n}\vert_{[-T,T]}\bigr)
     =\lim_n\Phi^n_{c_n}(x_n,y_n)\,,
     \end{equation}
     where $\Phi^n$ and $\Phi$ are the action potentials of $L+\psi_n$
     and of $L$ respectively. Write $\Delta:=\lim_n\Phi^n_{c_n}(x_n,y_n)$.
     We shall prove that $\Delta=\Phi_c(x,y)$, then~\eqref{cvs.1}
     becomes an equality and hence $x_v\vert_{[-T,T]}$ is semistatic.
     Since $T>0$ is arbitrary, then $v\in\S(L)$.
     
     Suppose that $\Phi_c(x,y)<\Delta - \e$, then there exists
     a curve $\eta:[0,S]\to M$ with $\eta(0)=x$, $\eta(S)=y$ and
     $A_{L+c}(\eta)<\Delta-\e$. Then
     \begin{equation}\label{cvs.2}
     \Phi^n_{c_n}(x_n,y_n)\le A_{L+\psi_n+c_n}(\eta)
     +\Phi^n_{c_n}(x,x_n)+\Phi^n_{c_n}(y,y_n).
     \end{equation}
     Fix a Riemannian metric on $M$.
     Using a speed~1 geodesic from $z_1\in M$ to $z_2\in M$, we get
     that
     $$
     \Phi^n_{c_n}(z_1,z_2)\le \bigl(\max_{(x,v)\in TM:\;|v|=1}
     |L(x,v)| +\max_{x\in M}
     |\psi_{n}(x)|  +c_n
     \bigr)\; d_M(z_1,z_2).
     $$
     Hence there exist $K>0$ such that for all $n$ sufficiently large we have
     $\Phi^n_{c_n}(z_1,z_2)\le K\; d_M(z_1,z_2)$.
     Letting $n\to\infty$ on equation~\eqref{cvs.2}, we get that
     $\lim_n\Phi^n_{c_n}(x_n,y_n)\le\Delta-\e$. This contradicts
     the definition of $\Delta$.
     

\qed

\begin{Lemma} If $\M^{0}(L)=\{\mu\}$ then $\hS(L)=\S(L)$.
\label{tres}
\end{Lemma}

\PROOF We show first that $\supp(\mu)$ is inside a static class.
    Since $\M^{0}(L)=\{\mu\}$, then $\mu$ is ergodic.
    In particular, $\mu$-almost every point has a dense orbit
    on $\supp(\mu)$. Let $v\in\supp(\mu)$ be such that it has
    a dense orbit on $\supp(\mu)$. Let $u,w\in\supp(\mu)$ and let
    $0<r_n<s_n<t_n$ be such that
    $\lim_n f_{r_n}(v)=u=\lim_n f_{t_n}(v)$ and
    $\lim_n f_{s_n}(v)=w$.
    Then
    \begin{align*}
    d_c\bigl(\pi f_{r_n}(v),\pi f_{t_n}(v)\bigr)
     &\le A_{L+c}\bigl(\pi f_{[r_n,s_n]}(v)\bigr)
         + A_{L+c}\bigl(\pi f_{[s_n,t_n]}(v)\bigr)
     \\
     &= A_{L+c}\bigl(\pi f_{r_n}(v),\pi f_{t_n}(v)\bigr)
     \\
     &=\Phi_c\bigl(\pi f_{r_n}(v),\pi f_{t_n}(v)\bigr),
    \end{align*}
    and
    \begin{align*}
    d_c(\pi u, \pi w)
       &=\lim_n d_c\bigl(\pi f_{r_n}(v),\pi f_{t_n}(v)\bigr)
       \\
       &\le \lim_n \Phi_c\bigl(\pi f_{r_n}(v),\pi f_{t_n}(v)\bigr)
       = \Phi_c(\pi u, \pi u) = 0,
    \end{align*}
 and hence $\supp(\mu)$ is inside a static class.

 Now let $v\in\S(L)$. For $S,T>0$ consider the probabilities
  $\nu_{ST}$ defined by
  $$
  \int g \; d\nu_{ST} = \tfrac{1}{S+T} \int_{-S}^T g(f_s(v))\;ds\;,
  $$
  for any $g:TM\to\re$ continuous. Since the $\om$~and~$\a$-limits
  of $v$ are in $\hS(L)$ and any weak limit of $\{\nu_{ST}\}_{S,T>0}$
  is invariant, then by Theorem IV in \cite{Ma,CDI},
  any weak* limit of $\nu_{ST}$ is minimizing and hence 
  it is $\mu$.
  
  Then $\a$-limit$(v)\subset\supp(\mu)$ and $\om$-limit$(v)\subset\supp(\mu)$.
  Let $u\in\a$-limit$(v)$, $w\in\om$-limit$(v)$ and
  $S_n,\,T_n\to+\infty$ such that $\lim_{n}f_{-S_n}(v)=u$ and
  $\lim_{n}f_{T_n}(v)=w$. For $s,t>0$ define
  \begin{equation}\label{cvs.3}
  \de(s,t)=A_{L+c}\bigl(\pi f_{[-s,t]}(v)\bigr)
  +\Phi_c(\pi f_t(v),f_{-s}(v)\bigr).
  \end{equation}
  Then the triangle inequality for $\Phi_c$ implies that
  $\de(-s,t)$ is increasing on $s>0$ and $t>0$. Also, since $v$ is
  semistatic, $\de(-s,t)=d_c\bigl(\pi f_{-s}(v),\pi f_t(v)\bigr)\ge 0$.
  But then, since $\supp(\mu)$ is inside a static class,
  $$
  \lim_n\de(-S_n,T_n)=d_c(\pi u, \pi w) = 0.
  $$
  Hence $\de(-s,t)\equiv 0$ for all $s,t>0$, and thus
  equation~\eqref{cvs.3} implies that $v\in\hS(L)$.

\qed

\begin{Lemma} Let
    $$
    \cG_2 := \{\, \psi\in C^\infty(M,\re)\,\vert\,
    \M^{0}(L+\psi)=\{\mu\} \text{ and } \hS(L+\psi)=\supp(\mu)\,\}.
    $$
    Then
    \begin{itemize}
    \item[(a)] $\cG_2$ is dense in $C^\infty(M,\re)$.
    \item[(b)] If $\psi_0\in\cG_2$, then 
               $\lim_{\psi\to\psi_0}d_{H}\big(\hS(L+\psi),\,\hS(L+\psi_0)\big)=0$
               where $d_{H}$ is the Hausdorff metric between compact subsets of $TM$.
    \item[(c)] If $\psi\in C^\infty(M,\re)$, $\mu_\psi\in\M^{0}(L+\psi)$
               and $\psi_0\in\cG_2$, then
               $$
               \lim_{\psi\to\psi_0}d_{H}\big(\supp(\mu_\psi)\;,\supp(\mu_{\psi_0})\big)=0
               $$.
               \end{itemize}     
\label{cuatro}
\end{Lemma}

\PROOF  Let us prove (a).
 By Theorem C in \cite{Ma1}, the set
     $$
     \cG_1:=\{\,\psi\in C^\infty(M,\re)\,\vert\,\#\M^{0}(L+\psi)=1\,\}
     $$
     is generic in $C^\infty(M,\re)$. We shall see that if
     $\psi_0\in\cG_1$, $\M^{0}(L+\psi_0)=\{\mu\}$ and $\psi_1\in
     C^\infty(M,\re)$ is such that $\psi_1\ge 0$ and
     $\{x:\;\psi_1(x)=0\}=\pi(\supp(\mu))$, then $\hS(L+\psi_0+\psi_1)=\supp(\mu)$
     and $\M^{0}(L+\psi_0+\psi_1)=\{\mu\}$. This implies that $\cG_2$
     is dense in $C^\infty(M,\re)$.
     
     Observe that $\mu\in\M^{0}(L+\psi_0+\psi_1)$ and hence
     $\supp(\mu)\subseteq\hS(L+\psi_0+\psi_1)$. If $v_0\in TM$ and
     $\pi(v_0)\notin\pi(\supp(\mu))$, we shall see that 
     $v_0\notin\hS(L+\psi_0+\psi_1)$. Then the Graph Property (cf.
     Theorem \ref{GraphProperty}) implies
     that $\hS(L+\psi_0+\psi_1)=\supp(\mu)$. Indeed, if
     $v_1=f_1^{L+\psi_0+\psi_1}(v_0)$, then
     \begin{align*}
     A&_{L+\psi_0+\psi_1+c}\left(\pi\,f_{[0,1]}^{L+\psi_0+\psi_1}(v_0)\right)
     +\Phi_c^{L+\psi_0+\psi_1}(\pi\,v_1\,,\,\pi\,v_0)
     \\
     &\ge
     \int_0^1\psi_1\left(\pi\,f_s^{L+\psi_0+\psi_1}(v_0)\right)\,ds
     +\Phi_c^{L+\psi_0}(\pi\,v_0\,,\,\pi\,v_1)
     +\Phi_c^{L+\psi_0}(\pi\,v_1\,,\,\pi\,v_0)
     \\
     &> d_c^{L+\psi_0}(\pi\,v_0,\pi\,v_1)\geq 0.
     \end{align*}
     Hence $v_0$ is not static.
     
     \medskip
     
     We now prove (b) and (c).
     From Lemmas \ref{dos} and \ref{tres}, if $\psi_0\in\cG_2$, then
     $\lim_{\psi\to\psi_0}\hS(L+\psi)\subseteq\hS(L+\psi_{0})=
     \supp(\mu_{\psi_0})$.
     On the other hand the continuity of the critical value ensures
     that the weak* limit of minimizing measures of $L+\psi$
     is minimizing for $L+\psi_0$ and hence 
     $\lim_{\psi\to\psi_0}\hS(L+\psi)\supseteq\hS(L+\psi_{0})$
     and thus $\lim_{\psi\to\psi_0}\hS(L+\psi)=\supp(\mu_{\psi_0})$.

     This implies that for any neighborhood $U$ of
     $\supp(\mu_{\psi_0})$ there is a neighborhood $\cV$ of
     $\psi_0$ such that $\hS(L+\psi)\subseteq U$ for all 
     $\psi\in\cV$. Let $d$ be the distance function of some Riemannian
     metric on $TM$.
     Using neighborhoods 
     $$U_\e:=\{\,z\in TM\,\vert\,d(z\,,\,\supp(\mu_{\psi_0}))<\e\,\},$$
     one gets that
     \begin{equation}\label{cvs.4}
     \lim_{\psi\to\psi_0}\sup_{z\in
     \La(L+\psi)}
     d\bigl(z\,,\,\supp(\mu_{\psi_0})\bigr)
     \le \lim_{\psi\to\psi_0}\;\sup_{z\in\hS(L+\psi)}\;
     d\bigl(z\,,\,\S(L+\psi_0)\bigr)=0.
     \end{equation}
     
     \medskip
     
     Given $\e>0$, let
     $\{z_1,\ldots,z_N\}\subset\supp(\mu_{\psi_0})$ be such that 
     $\supp(\mu_{\psi_0})\subset\bigcup_{i=1}^N B(z,\e_i)$, where
     $B(z,\e):=\{\,w\in TM\,\vert\, d(z,w)<\e\,\}$ and let
     $g_i:TM\to [0,1]$ be a non-constant positive continuous function
     with $\supp(g_i)\subseteq B(z_i,\e)$. Then $\int
     g_i\,d\mu_{\psi_0}>0$. The continuity of $c(L)$ implies that if
     $\psi\to\psi_0$ and $\mu_\psi\in\M^{0}(L+\psi)$ then
     $\mu_\psi\to\mu_{\psi_0}$ weakly*. Hence there is a neighborhood
     $\cV$ of $\psi_0$ such that if $\psi\in\cV$ and
     $\mu_\psi\in\M^{0}(L+\psi)$, then $\int g_i\, d\mu_\psi>0$ for all
     $i=1,\ldots,N$. Hence
     $$
     \lim_{\psi\to\psi_0}\sup_{z\in\supp(\mu_{\psi_0})}
     d\left(z\,,\La(L+\psi)\right)\le\e.
     $$
     Since this holds for any $\e>0$, then
     \begin{equation}\label{cvs.5}
     \lim_{\psi\to\psi_0}\sup_{z\in\hS(L+\psi_0)}
     d\big(z\,,\hS(L+\psi)\big)
     \le\lim_{\psi\to\psi_0}\sup_{z\in\,\supp(\mu_{\psi_0})}
     d\big(z\,,
     \La(L+\psi)\big)=0.
     \end{equation}
     From~\eqref{cvs.4} and~\eqref{cvs.5} we get that
     $$
     \lim_{\psi\to\psi_0} d_H\big(\hS(L+\psi),\hS(L+\psi_0)\big)
     =\lim_{\psi\to\psi_0}d_H\big(
     \La(L+\psi)\,,\,\supp(\mu_{\psi_0})\big)=0.
     $$
     
\qed


 To complete the proof of Theorem C we now show
that $\cG_2$ is generic.

We claim that the set 
     $$
     \cU(\e):=\{\,\psi\in C^\infty(M,\re)\,\vert\, 
     d_H\big(\hS(L+\psi)\;,
     \La(L+\psi)\big)<\e\,\}
     $$
     contains a neighborhood of $\cG_2$.
 
     This follows from parts (b) and (c) in Lemma \ref{cuatro} and
     the triangle inequality for
     the Hausdorff distance, i.e. using that
     $\hS(L+\psi_0)=\supp(\mu_{\psi_0})$ for $\psi_0\in\cG_2$, we have
     that 
     \begin{align*}
     d_H&\big(\hS(L+\psi),
     \La(L+\psi)\big)
     \le 
     \\
     &\le d_H\big(\hS(L+\psi),\hS(L+\psi_0)\big)
     +d_H\big(\supp(\mu_{\psi_0}),
     \La(L+\psi)\big).
     \end{align*}
     
     \medskip
     
     Since $\cG_2$ is dense,
     the set $\cU(\e)$ contains a open and dense set. Then
     $$
     \bigcap_{n>0}\cU(\tfrac 1n) =\big\{\,
     \psi\in C^\infty(M,\re)\,\vert\,
     \hS(L+\psi)=
     \La(L+\psi)\;\big\}
     $$
     is generic. Since $\cG_2=\cG_1\cap\big[\underset{n>0}\cap\cU(\tfrac
     1n)\,\big]$ and $\cG_1$ is generic, then $\cG_2$ is generic.
     
 
 
   \qed 
 









\section{Proof of Corollaries 2 and 3}

We need the following easy lemma.

\begin{Lemma}Let $M$ be a closed manifold with first Betti
number $b_{1}(M,\re)\geq 2$. Then if $A\subset M$ is a closed
submanifold diffeomorphic to $S^{1}$ and $U_{\varepsilon}$ denotes
the $\varepsilon$ neighborhood of $A$, we have that $H_{1}(M,U_{\varepsilon},\re)$
is non zero for all $\varepsilon$ sufficiently small.  \label{2}
\end{Lemma}

\PROOF Since $A$ is diffeomorphic to a circle, the singular homology 
of the pair $(M,U_{\varepsilon})$ coincides with the singular homology
of the pair $(M,A)$ and therefore
the vector space $H_{1}(M,U_{\varepsilon},\re)$ must have dimension $\geq b_{1}(M,\re)-1\geq 1$.

\qed

We recall the following generic property proved in \cite{CDII} and
\cite{Ma1} that we already mentioned in the introduction. 

\begin{Theorem} Given a
Lagrangian $L$ there exists a generic set ${\cal O}\subset C^{\infty}(M,\re)$
such that if $\psi\in{\cal O}$ the Lagrangian $L+\psi$ has a unique
minimizing measure in ${\cal M}^{0}(L+\psi)$ and this measure
is uniquely ergodic. When this measure is supported on a periodic orbit, this
orbit is hyperbolic and {\bf if} the stable and unstable manifolds intersect, they must do it transversally.   \label{teo}
\end{Theorem}

It is conjectured in \cite{Ma} that the unique
minimizing measure in ${\cal M}^{0}(L+\psi)$ is always supported on a periodic orbit.

Observe now that if we combine Corollary 1, Lemma \ref{2} and Theorem \ref{teo} we obtain Corollary 2.

\qed

To prove Corollary 3 we need the following lemma.
A proof can be found in \cite[Proposition 8]{Fa3}.

\begin{Lemma}If $L$ is a symmetric Lagrangian, then
\[c(L)=-\inf_{x\in M}L(x,0),\]
and
\[\La(L)=\hS(L)=\{(x,0):\;L(x,0)=-c(L)\}.\]
Moreover, the ergodic minimizing measures are the Dirac measures
concentrated on the fixed points $(x,0)$ of the
Euler-Lagrange flow with $L(x,0)=-c(L)$.
\label{3}
\end{Lemma}

Finally observe that if we combine Corollary 1, Lemma \ref{3} and Theorem \ref{teo} we obtain Corollary 3.





\qed





 





\end{document}
\end













